



ACCEPTED MANUSCRIPT • OPEN ACCESS

A survey on the Ulam-Hyers stability of fractional-order differential equations

To cite this article before publication: Matap Shankar et al 2025 J. Phys. A: Math. Theor. in press https://doi.org/10.1088/1751-8121/ae189e

Manuscript version: Accepted Manuscript

Accepted Manuscript is "the version of the article accepted for publication including all changes made as a result of the peer review process, and which may also include the addition to the article by IOP Publishing of a header, an article ID, a cover sheet and/or an 'Accepted Manuscript' watermark, but excluding any other editing, typesetting or other changes made by IOP Publishing and/or its licensors"

This Accepted Manuscript is © 2025 The Author(s). Published by IOP Publishing Ltd.

As the version of Record of this article is going to be / has been published on a gold open access basis under a CC BY 4.0 licence, this Accepted Manuscript is available for reuse under a CC BY 4.0 licence immediately.

Everyone is permitted to use all or part of the original content in this article, provided that they adhere to all the terms of the licence https://creativecommons.org/licences/by/4.0

Although reasonable endeavours have been taken to obtain all necessary permissions from third parties to include their copyrighted content within this article, their full citation and copyright line may not be present in this Accepted Manuscript version. Before using any content from this article, please refer to the Version of Record on IOPscience once published for full citation and copyright details, as permissions may be required. All third party content is fully copyright protected and is not published on a gold open access basis under a CC BY licence, unless that is specifically stated in the figure caption in the Version of Record.

View the <u>article online</u> for updates and enhancements.

A survey on the Ulam-Hyers stability of fractional-order differential equations

Matap Shankar $^{\dagger},$ Ralf Metzler $^{\sharp},$ and Changpin Li †

- † Department of Mathematics, Shanghai University, Shanghai 200444, China
- # University of Potsdam, Institute for Physics & Astronomy, 14476 Potsdam, Germany and Asia Pacific Centre for Theoretical Physics, Pohang 37673, Republic of Korea

Abstract. Stability theory plays a central role in the analysis of the behaviour of a solution of a real-order differential equations. In the literature, various stability concepts were introduced from the application point of view. The most popular ones are the Lyapunov and the Ulam-Hyers stability. Here, we first we establish a relation between the Lyapunov and Ulam-Hyers stability concepts for a dynamical system and prove that the concept of Ulam-Hyers is more general than that of Lyapunov. Second, we present a brief overview of recent developments in the Ulam-Hyers stability analysis of fractional-order differential equations (FDEs). These equations include linear FDEs, non-linear FDEs, delay FDEs, fractional-order boundary value problems and impulsive FDEs.

1. Introduction

In 1940, Ulam introduced the following problem regarding the stability of the group homomorphism in a metric group (G, \star, d_{\star}) [70] where \star represents the operation in the given group G. For a given function $g: G \to G$ and a positive number ε satisfying

$$d_{\star}(g(e_1 \star e_2), g(e_1) \star g(e_2)) \le \varepsilon, \quad \forall e_1, e_2 \in G, \tag{1.1}$$

does there exist a positive constant L and a homomorphism $h: G \to G$ of the group (i.e., $h(e_1 \star e_2) = h(e_1) \star h(e_2)$) with the property

$$d_{\star}(g(e_1), h(e_1)) \le L \,\varepsilon, \quad \forall \, e_1 \in G? \tag{1.2}$$

If such a constant L exists that satisfies equation (1.2), then we say that the equation of the homomorphism

$$h(e_1 \star e_2) = h(e_1) \star h(e_2)$$
 (1.3)

on the metric group is stable in the Ulam sense. A year later, Hyer provided an answer to Ulam's problem for additive functions on Banach spaces as follows [31]:

For real Banach spaces $(X, ||.||_X)$ and $(Y, ||.||_Y)$; given a function $g: X \to Y$ and a positive number ε satisfying

$$||g(x+y) - g(x) - g(y)||_{Y} \le \varepsilon \quad \forall x, y \in X,$$
(1.4)

there exists a unique additive function $f: X \to Y$ i.e., f(x+y) = f(x) + f(y) satisfying

$$||g(x) - f(x)||_{Y} \le \varepsilon \quad \forall x \in X.$$
 (1.5)

Such a type of stability concept is known as Ulam-Hyers stability (also Hyers-Ulam or Ulam stability). Generally speaking, a functional equation is said to be Ulam-Hyers stable, if for every solution of the perturbed equation (called approximate solution), there exists a solution of the equation (exact solution) near to it. After the Hyers result on stability many researchers have extended the concept of Ulam-Hyers stability to other functional equations, we refer the reader to [12, 13, 32, 33, 55, 58].

In 1978, Rassias [56] generalised the stability result of Hyers, an approach known as the Ulam-Hyers-Rassias stability. The stability result of Rassias is contained in the following theorem.

Theorem 1.1. [56] Consider X, Y to be two Banach spaces, and let $g: X \to Y$ be a mapping such that f(tx) is continuous in t for each fixed x. Assume that there exists $\theta > 0$ and $p \in [0,1)$ such that

$$||g(x+y) - g(x) - g(y)||_{Y} \le \theta(||x||_{X}^{p} + ||y||_{X}^{p}), \ \forall x, y \in X.$$
(1.6)

Then, there exists a unique linear mapping $f: X \to Y$ such that

$$||g(x) - f(x)||_{Y} \le \left(\frac{2\theta}{2 - 2^{p}}\right) ||x||_{X}^{p}, \ \forall x \in X.$$
 (1.7)

The positive number L in Ulam's problem is called a Ulam-Hyers constant corresponding to the considered equation (1.3). Let us denote L_M the infimum of all Ulam-Hyers constants L. In general L_M may not be a Ulam-Hyers constant for a given functional equation (see [26]). In case when L_M is an Ulam-Hyers constant it is called the best Ulam-Hyers constant. The best Ulam-Hyers constant has a major significance in the analysis of the Ulam-Hyers stability of a given functional equation. This is, because one can not only estimate the possible minimum distance between the approximate and the exact solution of the functional equation, but, at the same time, also get away with searching for other possible Ulam-Hyers constants by different approaches. For instance, Onitsuka and Shoji [52] studied the Ulam-Hyers stability of the first-order linear differential equation

$$x'(t) - ax(t) = 0, \quad \forall t \in \mathbb{R}, \tag{1.8}$$

where a is a non-zero real number and evaluated the best Ulam-Hyers constant $L_M = 1/|a|$. Now, there is no interest in studying the Ulam-Hyers stability of the above problem (1.8) because whatever different approach we pursue, it will always end up with a Ulam-Hyers constant $L \geq L_M$ and thus will be a bad estimation of the bound for the distance between the approximate and the exact solution of equation (1.8). Finding the best Ulam-Hyers constant, if exists, for a given functional equations is a challenging task, except for some linear functional equations—finding the best Ulam-Hyers constant for non-linear functional equations is still an open problem. However, if the given functional equation does not possess the best Ulam-Hyers constant or one is unable to find the best Ulam-Hyers constant, then one can apply different approaches to improve the estimation for the Ulam-Hyers constant L.

In a number of practical problems, integral-order derivatives and differential equations do not convey the full picture of the situation. In many complex systems, the use of the concept of fractional-order calculus [15,24,29,36,42,44,47–49,51,54,65] or fractional Brownian motion [6,9,25,35,43,46,71] offers new ways to tackle these problems Fractional-order differintegral operators incorporate long-range memory effects in terms of power-law kernels. For instance, fractional-order, linear dynamic equations can be derived from continuous time random walk processes based on scale-free sojourn time densities in the hydrodynamic limit [47,48]. Fractional differential equations (FDEs) of different types (e.g., linear and non-linear) will be considered in the following.

The study of real-order differential equations is mainly divided into two parts: quantitative and qualitative theory. The qualitative theory is considerably more effective than the quantitative theory in the analysis of real-order differential equations. The analysis of the qualitative properties of such differential equations is of high interest since differential equations arise in nearly all disciplines of science, medicine, engineering, economics, demography, geophysics, and biocenology. The qualitative theory deals with diverse topics along with their physical existence, such as stability, asymptotic stability, periodic orbit, limit cycles, and chaos. Stability theory is one of the oldest and most effective concepts to analyse the dynamics of differential equations and to design control in numerous complex engineering problems. In the literature, depending upon the requirement to handle the mathematical difficulty and from an application point of view, various stability concepts have been introduced to analyse the behaviour of some of the physical states connected to the real-order differential equations. As mentioned, these include the Ulam-Hyers, Ulam-Hyers-Rassias, and Lyapunov stability, inter alia. In the case

of integer-order dynamical systems the Lyapunov stability theory is a very mature subject and has a very rich mathematical foundation [34,72]. However, over the last few decades, a good amount of research has been carried out on the applications of Lyapunov stability for fractional-order dynamical systems to more realistic problems in science and engineering [7,67,68]. For detailed discussion on Lyapunov stability results for FDEs we refer to the survey paper [40]. In what follows, we present different methods in a survey on the Ulam-Hyers stability results for various classes of FDEs.

The remainder of this paper is organised as follows. Section 2 introduces basic notations, definitions, and preliminary results for FDE stability analysis. Section 3 presents the concept of strong Ulam-Hyers stability and gives its relationship with Lyapunov stability in dynamical systems. Section 4 discusses Ulam-Hyers stability results of linear FDEs, while section 5 examines the nonlinear case. Section 6 deals with the Ulam-Hyers stability conditions of FDEs with delay. In section 7, we present Ulam-Hyers stability conditions of fractional boundary value problems. In section 8, we show the Ulam-Hyers stability results of fractional impulsive differential equations are described. Finally, section 9 displays concluding remarks.

2. Preliminaries

In this section, we introduce some definitions and results, which will be used throughout this work.

Let $\mathbb{C}, \mathbb{R}, \mathbb{N}$, and \mathbb{R}_+ , denote the set of complex numbers, the set of real numbers, the set of natural numbers, and the set of positive real numbers, respectively. Furthermore, let \mathbb{R}^d $(d \in \mathbb{N})$ denote the d-dimensional Euclidean space.

Definition 2.1. [54] The Euler gamma function $\Gamma(z)$ is defined by the integral

$$\Gamma(z) = \int_0^\infty e^{-\tau} \tau^{z-1} d\tau, \qquad (2.1)$$

which converges in the right half of the complex plane, Re(z) > 0.

Definition 2.2. [54] The classical Mittag-Leffler function is the generalisation of the exponential function. The one-parameter Mittag-Leffler function E_{α} and two-

parameter (generalised) Mittag-Leffler function $E_{\alpha,\beta}$ are defined, respectively, by

$$E_{\alpha}(z) = \sum_{k=0}^{\infty} \frac{z^k}{\Gamma(\alpha k + 1)}, \quad z \in \mathbb{C}, \operatorname{Re}(\alpha) > 0,$$
(2.2)

$$E_{\alpha,\beta}(z) = \sum_{k=0}^{\infty} \frac{z^k}{\Gamma(\alpha k + \beta)}, \quad z \in \mathbb{C}, \operatorname{Re}(\alpha) > 0, \operatorname{Re}(\beta) > 0.$$
 (2.3)

Lemma 2.1. [75] For any $\lambda \geq 0$ and $t \in [0, \infty)$

$$E_{\alpha}(-\lambda t^{\alpha}) \le 1, \quad E_{\alpha,\alpha}(-\lambda t^{\alpha}) \le \frac{1}{\Gamma(\alpha)}.$$
 (2.4)

Definition 2.3. [54] The Riemann-Liouville fractional integral $_{RL}I_{a,t}^{\alpha}$ of order $\alpha > 0$ of a given function v(t) is defined by

$$_{RL}I_{a,t}^{\alpha}v(t) = \frac{1}{\Gamma(\alpha)} \int_{a}^{t} (t-\tau)^{\alpha-1}v(\tau) d\varsigma, \quad t > a.$$
 (2.5)

Definition 2.4. [54] The Riemann-Liouville fractional derivative $_{RL}D_{a,t}^{\alpha}$ of order $\alpha > 0$ of a given function v(t) is defined by

$$_{RL}D_{a,t}^{\alpha}v(t) = \begin{cases} \frac{1}{\Gamma(n-\alpha)} \frac{d^n}{dt^n} \int_a^t (t-\tau)^{n-\alpha-1} v(\tau) d\tau, & n-1 < \alpha < n, \ t > a, \\ v^{(n)}(t), & \alpha = n, \end{cases}$$
(2.6)

where $n = \lceil \alpha \rceil$ is a positive integer; here $\lceil \cdot \rceil$ denotes the ceiling function.

Definition 2.5. [54] The Caputo fractional derivative ${}_{C}D^{\alpha}_{a,t}$ of order $\alpha > 0$ of a given function v(t) is defined by

$${}_{C}\mathrm{D}_{a,t}^{\alpha}v(t) = \begin{cases} \frac{1}{\Gamma(n-\alpha)} \int_{a}^{t} (t-\tau)^{n-\alpha-1} v^{(n)}(\tau) d\tau, & n-1 < \alpha < n, \ t > a, \\ v^{(n)}(t), & \alpha = n. \end{cases}$$
(2.7)

where $n = \lceil \alpha \rceil$ is a positive integer, and $v^{(n)}(t) = \frac{d^n v}{dt^n}$.

Definition 2.6. [19] For $\rho \in (0,1]$ and $\alpha > 0$. 0. The generalised proportional fractional integral $\mathscr{I}_{a,t}^{\alpha,\rho}$ of order $\alpha > 0$ of a given function v(t) is defined by

$$\mathscr{I}_{a,t}^{\alpha,\rho}v(t) = \frac{1}{\rho^{\alpha}\Gamma(\alpha)} \int_{a}^{t} e^{\frac{\rho-1}{\rho}(t-\tau)} (t-\tau)^{\alpha-1} v(\tau) d\tau, \quad t > a.$$
 (2.8)

Definition 2.7. [19] For $\rho \in (0,1]$ and $0 < \alpha \le 1$. 0. The generalised proportional fractional derivative of Riemann-Liouville type $_{RL}D_{a,t}^{\alpha,\rho}$ of order α of a given function v(t) is defined by

$${}_{RL}D_{a,t}^{\alpha,\rho}v(t) = \begin{cases} \frac{1}{\rho^{1-\alpha}\Gamma(1-\alpha)}D_t^{1,\rho}\left(\int_a^t e^{\frac{\rho-1}{\rho}(t-\tau)}(t-\tau)^{-\alpha}v(\tau)\,d\tau\right), & 0 < \alpha < 1, \ t > a, \\ \mathcal{D}_t^{1,\rho}v(t), & \alpha = 1, \end{cases}$$
(2.9)

where $\mathcal{D}_{t}^{1,\rho}v(t) = (1-\rho)v(t) + \rho v'(t)$.

Integral inequalities play an important role in establishing the stability conditions for FDEs. In the literature various integral inequalities have been introduced, including the Grönwall inequality, generalised Grönwall inequality, Henry-Grönwall inequality, etc. Below, we state those integral inequalities which are essential for obtaining the results on the Ulam-Hyers stability.

Theorem 2.1. [34] (Grönwall's Inequality) Suppose that u(t) and v(t) are continuous real-valued functions defined on $0 \le t < T$ ($T \le +\infty$) with $u(t) \ge 0$. Assume that u and v satisfy

$$u(t) \le k_1 + k_2 \int_0^t v(\tau) u(\tau) d\tau$$

on $0 \le t < T$, where k_1 and k_2 are constants with $k_2 \ge 0$. Then,

$$u(t) \le k_1 \exp(k_2 \int_0^t v(\tau) d\tau), \quad \forall t \in [0, T).$$

Theorem 2.2. [83] (Henry-Grönwall inequality) Suppose that $\alpha > 0$, $g \in C([0,T),\mathbb{R}_+)$ is a non-decreasing function and $a : [0,T) \to \mathbb{R}_+$ is a locally integrable non-decreasing function; moreover, suppose that u(t) is locally integrable non-negative with

$$u(t) \le a(t) + g(t) \int_0^t (t - \tau)^{\alpha - 1} u(\tau) d\tau$$

on $0 \le t < T$. Then,

$$u(t) \le a(t)E_{\alpha}(g(t)\Gamma(\alpha)t^{\alpha}), \quad \forall t \in [0, T).$$

Another important element in the analysis of FDEs are fixed point theorems, without which it is very difficult (actually almost impossible) to study the existence and uniqueness of non-linear FDEs. Fixed point theorems are nowadays the most

widely used tool for studying the Ulam-Hyers stability of FDEs. For details on the application of fixed point theorems to Ulam-Hyers stability of functional equations we refer to the survey paper [16] Details on fixed point theorems are described in the monograph [53]. Below, we state those fixed point theorems which are essential for obtaining the results on the Ulam-Hyers stability.

Theorem 2.3. [53] (Banach Fixed Point Theorem) Let (M, d_M) be a complete metric space. Let $\Upsilon: M \to M$ be a contraction map with the Lipschitz constant L < 1. If there exists a non-negative integer k such that $d_M(\Upsilon^{k+1}y, \Upsilon^k y) < +\infty$ for some $y \in M$, then

- (i) the sequence $\{\Upsilon^n y\}$ converges to a fixed point x^* of Υ ,
- (ii) x^* is the unique fixed point of Υ in $M^* = \{z \in M \mid d_M(\Upsilon^k y, z) < \infty\}$,
- (iii) if $z \in M^*$, then $d_M(z, x^*) \leq \frac{1}{1-L} d_M(\Upsilon z, z)$.

For a given FDE, the Banach fixed point theorem is not only helpful to prove the Ulam-Hyers stability of the problem but it also help to estimate the Ulam-Hyers constant.

Theorem 2.4. [53] (Krasnoselskii's Fixed Point Theorem) Let $N(\neq \emptyset)$ be a closed, convex subset of a Banach space M and $\Upsilon_1, \Upsilon_2 : M \to M$ be two operators satisfying $(i)\Upsilon_1 u + \Upsilon_2 v \in N$, whenever $u, v \in N$,

- $(ii)\Upsilon_1$ is continuous and compact,
- $(iii)\Upsilon_2$ is a contraction operator.

Then there exists $w^* \in N$ such that $w^* = \Upsilon_1 w^* + \Upsilon_2 w^*$.

Theorem 2.5. [53] Let X be a Banach space and $\Upsilon: X \to X$ a completely continuous operator. If the set

$$G(\Upsilon) = \{x \in X : x = \lambda \Upsilon(x), \text{ for some } \lambda \in [0, 1]\}$$

is bounded, then Υ has a fixed point.

Now, we introduce the concept of the Ulam-Hyers stability for FDEs. Consider the general FDE:

$$\mathcal{F}(q(t), u(t), \mathcal{D}^{\alpha_1} u(t), \dots, \mathcal{D}^{\alpha_n} u(t)) = 0, \quad \forall t \in [a, b], \ -\infty \le a < b \le +\infty, \quad (2.10)$$

where, $u:[a,b]\to X$, $((X,\|.\|_X)$ is a norm linear space), D^{α_i} $i=1,\ldots n$ are fractional-order differential operators with $0\leq \alpha_1\leq \cdots \leq \alpha_n$, and $q:[a,b]\to X$ is a given function.

Given $\varepsilon > 0$, and $\varphi : [a, b] \to \mathbb{R}_+$. Suppose $v : [a, b] \to X$ satisfy one of the following FDEs:

$$\|\mathcal{F}(q(t), v(t), D^{\alpha_1}v(t), \dots, D^{\alpha_n}v(t))\| \le \varepsilon, \quad \forall t \in [a, b],$$
(2.11)

$$\|\mathcal{F}(q(t), v(t), \mathbf{D}^{\alpha_1} v(t), \dots, \mathbf{D}^{\alpha_n} v(t))\| \le \varphi(t), \quad \forall t \in [a, b],$$
 (2.12)

Definition 2.8. (Ulam-Hyers stability) The FDE (2.10) is Ulam-Hyers stable if there exists a positive constant L > 0, such that for each $\varepsilon > 0$ and for each solution v of equation (2.11) there exists an exact solution u of (2.10) with

$$||v(t) - u(t)|| \le L \varepsilon, \quad \forall t \in [a, b]. \tag{2.13}$$

such L is often called Ulam-Hyers constant, and it is independent of ε .

Definition 2.9. (Ulam-Hyers-Rassias stability) The FDE (2.10) is Ulam-Hyers-Rassias stable, if there exists a positive constant L > 0 such that for each solution v of inequation (2.12) there exists a solution v of (2.10) with:

$$||v(t) - u(t)|| \le L\varphi(t), \quad \forall t \in [a, b].$$
 (2.14)

2.1. Useful tools to analyse the Ulam-Hyers stability

As a summary of this section we list the three main classes of tools to analyse the Ulam-Hyers stability of FDEs.

- (i) Fixed point approach: This approach applies to non-linear problems. The most frequently used fixed point theorems to establish the Ulam-Hyers stability for FDEs are the Banach fixed point theorem, the non-linear alternative of the Leray-Schauder type, Krasnoselskii's and Schauder's fixed point theorems, etc.
- (ii) Integral transforms approach: When a given problem is linear, then integral transform approach will be a good choice to analyse the Ulam-Hyers stability of the problem. The most commonly used integral transforms are the Laplace, Mellin, Sumudu, and Fourier transforms, etc.
- (iii) Functional inequalities approach: This plays a vital role for estimating the Ulam-Hyers constant. For some cases this approach helps directly to establish the Ulam-Hyers stability results but in most cases it is used along with the fixed point or integral transform approach. The most frequently used functional inequalities are Grönwall's inequality, the generalised Grönwall's inequality, Henry-Grönwall's inequality, the comparison theorem of differential equations, integral inequalities, etc.

‡ An inequation denotes a mathematical relation that is either an inequality or a "not equal to" relation between two values.

3. Relation between Lyapunov and Ulam-Hyers stability

In general, the Lyapunov and Ulam-Hyers stabilities are independent concepts. For instance, we can have the notion of Ulam-Hyers stability for any functional equation, but the concept of Lyapunov stability is confined to an equation representing the dynamical system. If we turn to dynamic systems, the Lyapunov stability deals in studying the behaviour of the solutions of a dynamical system near the equilibrium points of the system, whereas the Ulam-Hyers stability mainly applies to finding an exact solution near an approximate solution of the system. In this section, we approach the connection between Lyapunov's and Ulam-Hyers' stability concepts in the case of dynamical systems.

Consider the fractional differential equation in the dynamical systems

$$\begin{split} {}_C\mathrm{D}^{\alpha}_{0,t}\mathbf{x}(t) &= \mathbf{F}(t,\mathbf{x}), \quad t>0,\\ \mathbf{x}(0) &= \mathbf{x}_0 \end{split}$$
 where $0<\alpha<1,\,\mathbf{x}_0\in\Omega\subseteq\mathbb{R}^d,\,\mathrm{and}\,\mathbf{F}:[0,+\infty)\times\Omega\to\mathbb{R}^d$ is a continuous function.

Definition 3.1. A vector $\overline{\mathbf{x}} \in \Omega$ is said to be an equilibrium point of the first differential equation of the system (3.1) if $\mathbf{F}(t, \overline{\mathbf{x}}) = 0$, for all $t \geq 0$. Define the set $E \subseteq \Omega$ the collection of all such equilibrium points.

Denote $\mathbf{x}(t, \mathbf{x}_0)$ as the solution of the above differential equation (3.1) starting at an initial point $\mathbf{x}(0) = \mathbf{x}_0$.

Definition 3.2. [37] (Lyapunov stability) The equilibrium point $\overline{\mathbf{x}} \in E$ of the system (3.1) is Lyapunov stable, if for every $\varepsilon > 0$ there exists $\delta = \delta(\varepsilon) > 0$ such that

$$\|\mathbf{x}_0 - \overline{\mathbf{x}}\| < \delta \Rightarrow \|\mathbf{x}(t, \mathbf{x}_0) - \overline{\mathbf{x}}\| \le \varepsilon, \quad \forall t \ge 0,$$
 (3.2)

where $\|\cdot\|$ denotes a norm on \mathbb{R}^d . In other words, $\overline{\mathbf{x}} \in E$ is Lyapunov stable, if given any $\varepsilon > 0$ there exists a neighbourhood $N_{\delta}(\overline{\mathbf{x}})$ for some $\delta > 0$ such that for each $\mathbf{x}_0 \in N_{\delta}(\overline{\mathbf{x}})$, the solution $\mathbf{x}(t, \mathbf{x}_0)$ satisfies $\|\mathbf{x}(t, \mathbf{x}_0) - \overline{\mathbf{x}}\| \le \varepsilon$, for all $t \ge 0$.

Definition 3.3. (Ulam-Hyers stability) The differential equation (3.1) is said to be Ulam-Hyers stable if there exists a constant L>0, such that for every $\varepsilon>0$ and any function $\mathbf{y}:[0,\infty)\to\mathbb{R}^d$ satisfying

$$\left\| {}_{C}\mathrm{D}_{0,t}^{\alpha}\mathbf{y}(t) - \mathbf{F}(t,\mathbf{y}) \right\| \le \varepsilon, \quad \forall \, t \ge 0$$
 (3.3)

there exists an exact solution $\mathbf{x}:[0,\infty)\to\mathbb{R}^d$ of the differential equation (3.1) such that

$$\|\mathbf{y}(t) - \mathbf{x}(t)\| \le L\varepsilon, \quad \forall t \ge 0.$$
 (3.4)

Remark 3.1. In Definition 3.3, the existence of at least one solution $\mathbf{x} : [0, \infty) \to \mathbb{R}^d$ to the differential equation (3.1) that satisfies the inequality (3.4) is sufficient for Ulam-Hyers stability. One obvious such solution can be obtained by choosing an initial value $\mathbf{x}(0)$ depending on \mathbf{y} . In the literature, many authors set $\mathbf{x}(0) = \mathbf{y}(0)$. However, in some cases, there may exist multiple such solutions or even a whole family of solutions in a neighbourhood of $\mathbf{y}(0)$. This fact is demonstrated in the following theorem, established by Onitsuka and Shoji [52] for integer-order differential equations.

Theorem 3.1. [52] Consider the homogeneous linear differential equation

$$x'(t) = ax(t), \quad t \in I, \tag{3.5}$$

where I is a nonempty open interval of \mathbb{R} and a is a non-zero real number. Let $\varepsilon > 0$ be a given arbitrary constant. Suppose that a differentiable function $y: I \to \mathbb{R}$ satisfies

$$|y'(t) - ax(t)| \le \varepsilon, \quad \forall t \in I.$$
 (3.6)

Then one of the following holds:

- (i) if a > 0 and $\sup I$ exists, then $\lim_{t \to \tau 0} y(t)$ exists where $\tau = \sup I$, and any solution x(t) of (3.5) with $|\lim_{t \to \tau 0} y(t) x(\tau)| < \varepsilon/a$ satisfies that $|y(t) x(t)| < \varepsilon/a$ for all $t \in I$;
- (ii) if a > 0 and $\sup I$ does not exist, then $\lim_{t \to \infty} y(t)e^{-at}$ exists, and there exists exactly one solution $x(t) = \left(\lim_{t \to \infty} y(t)e^{-at}\right)e^{at}$ of (3.5) such that $|y(t) x(t)| < \varepsilon/a$ for all $t \in I$;
- (iii) if a < 0 and $\inf I$ exists, then $\lim_{t \to \sigma + 0} y(t)$ exists where $\sigma = \inf I$, and any solution x(t) of (3.5) with $|\lim_{t \to \sigma + 0} y(t) x(\sigma)| < \varepsilon/|a|$ satisfies that $|y(t) x(t)| < \varepsilon/|a|$ for all $t \in I$;
- (iv) if a < 0 and $\inf I$ does not exist, then $\lim_{t \to -\infty} y(t)e^{-at}$ exists, and there exists exactly one solution $x(t) = \left(\lim_{t \to -\infty} y(t)e^{-at}\right)e^{at}$ of (3.5) such that $|y(t) x(t)| < \varepsilon/|a|$ for all $t \in I$.

Moreover, they show that for a = 0, the differential equation (3.5) is not Ulam-Hyers stable.

Example 3.1. Consider the simple integer-order differential equation $(\alpha = 1)$:

$$x'(t) = ax(t), \quad a \in \mathbb{R} \setminus \{0\}, \ t \ge 0, \tag{3.7}$$

with the initial condition

$$x(0) = x_0. (3.8)$$

Let $\varepsilon > 0$ be a given arbitrary number and suppose that the differentiable function $y:[0,\infty) \to \mathbb{R}$ satisfies

$$|y'(t) - ay(t)| \le \varepsilon, \quad \forall t \ge 0. \tag{3.9}$$

If a > 0, then by (ii) assertion of Theorem 3.1, $\lim_{t \to \infty} y(t)e^{-at}$ exists. Denote this limit by b. Then, the IVP with initial condition x(0) = b has a unique solution $x(t) = be^{at}$ satisfying

$$|y(t) - x(t)| < \varepsilon/a. \tag{3.10}$$

In other words, the differential equation (3.7) is Ulam-Hyers stable with Ulam-Hyers constant given by L = 1/a and the solution x(t) which satisfied the inequality (3.10) is unique.

Inequality (3.9) implies

$$y'(t) = ay(t) + h(t),$$
 (3.11)

where $h:[0,\infty)\to\mathbb{R}$ and $|h(t)|\leq \varepsilon$ for all $t\geq 0$. Therefore, from (3.11) and (3.7), we obtain

$$y'(t) - x'(t) = a(y - x) + h(t).$$

The solution of the above equation is given by

$$y(t) - x(t) = (y(0) - x(0))e^{at} + \int_0^t h(s)e^{(t-s)} ds,$$

which implies

$$|y(t) - x(t)| \le |y(0) - x_0|e^{at} + \varepsilon \int_0^t e^{a(t-s)} ds.$$
 (3.12)

If a < 0, then $e^{at} \le 1$ for all $t \ge 0$. Choose an initial condition x_0 such that

$$|y(0) - x_0| \le \varepsilon. \tag{3.13}$$

Using these facts in (3.12), we get

$$|y(t) - x(t)| \le \left(1 + \frac{1}{|a|}\right)\varepsilon$$
 (3.14)

where x(t) is the solution of the IVP (3.7) with initial condition $x(0) = x_0$, and x_0 satisfying the inequality (3.13). In other words, the differential equation (3.7) is Ulam-Hyers stable with Ulam-Hyers constant $L = 1 + \frac{1}{|a|}$. Moreover, there exists a family of solutions to the IVP (3.7) that satisfy the inequality (3.14), where the initial values x_0 belong to an ε -neighbourhood of y(0) (i.e., $x_0 \in N_{\varepsilon}(y(0))$).

This example motivates to define a special class of Ulam-Hyers stability, in order to distinguish between two cases:

- (i) The discrete case, where the initial value \mathbf{x}_0 of the solution $\mathbf{x}(t)$ satisfying inequality (3.4) is distributed discretely (as in the case of a > 0).
- (ii) The continuous case, where the initial value \mathbf{x}_0 of the solution $\mathbf{x}(t)$ satisfying inequality (3.4) is distributed in some neighbourhood $N(\mathbf{y}(0))$ of $\mathbf{y}(0)$ (as in the case of a < 0).

A comparison between Ulam-Hyers and Lyapunov stability makes sense only when the initial condition \mathbf{x}_0 lies in a neighbourhood of the equilibrium point. Therefore, we introduce the following definition, which we call strong Ulam-Hyers stability.

Definition 3.4. (Strong Ulam-Hyers stability) The fractional differential equation (3.1) is said to be strongly Ulam-Hyers stable if there exists a constant L > 0 such that, for every $\varepsilon > 0$ and any vector-valued function $\mathbf{y} : [0, \infty) \to \Omega$ satisfying

$$\|_C D_{0,t}^{\alpha} \mathbf{y}(t) - \mathbf{F}(t, \mathbf{y}(t))\| \leqslant \varepsilon, \quad \forall t \geqslant 0,$$
 (3.15)

there exists $\delta = \delta(\varepsilon) > 0$ such that

$$\|\mathbf{x}_0 - \mathbf{y}(0)\| < \delta \implies \|\mathbf{y}(t) - \mathbf{x}(t)\| \le L\varepsilon, \quad \forall t \ge 0,$$
 (3.16)

where $\mathbf{x}(t)$ is the solution of the equation (3.1) subject to the initial condition $\mathbf{x}(0) = \mathbf{x}_0$, with \mathbf{x}_0 satisfying the first inequality in (3.16).

Now, we proceed to introduce a theorem that connects Lyapunov and strong Ulam-Hyers stability for the fractional dynamical system (3.1).

Theorem 3.2. For $0 < \alpha < 1$, the strong Ulam-Hyers stability of the fractional differential equation (3.1) implies Lyapunov stability of its equilibrium point.

Proof. Let $\varepsilon > 0$ be an arbitrary given number. Suppose $\overline{\mathbf{x}} \in E$ is an equilibrium point of the fractional differential equation (3.1). Then,

$$_{C}\mathrm{D}_{0,t}^{\alpha}\overline{\mathbf{x}} - \mathbf{F}(t,\overline{\mathbf{x}}) = 0, \quad \forall t \geqslant 0.$$
 (3.17)

Given that the fractional differential equation (3.1) is strongly Ulam-Hyers stable, there exists a constant L independent of ε . If we take $\varepsilon' = \varepsilon/L > 0$, then we observe that

$$||_C D_{0,t}^{\alpha} \overline{\mathbf{x}} - \mathbf{g}(t, \overline{\mathbf{x}})|| \leq \varepsilon', \quad \forall t \geqslant 0.$$

This inequality corresponds to equation (3.15) under the substitution $\mathbf{y} = \overline{\mathbf{x}}$. By Definition 3.4 (Strong Ulam-Hyers stability), there exists $\delta = \delta(\varepsilon') = \delta(\varepsilon/L) > 0$ such that

$$\|\mathbf{x}_0 - \overline{\mathbf{x}}\| < \delta \implies \|\mathbf{x}(t) - \overline{\mathbf{x}}\| \leqslant L\varepsilon' = \varepsilon, \quad \forall t \geqslant 0.$$
 (3.18)

This ends the proof.

Remark 3.2. The inverse of Theorem 3.2 remains an open problem. While Ulam-Hyers stability implies Lyapunov stability in the continuous case, the discrete case—when the existence of a solution $\mathbf{x}(t)$ satisfying the Ulam-Hyers inequality (3.4) is determined by a discrete distribution of the initial condition \mathbf{x}_0 —presents advantages over Lyapunov stability. For instance, in the above example for a > 0, the system (3.7) exhibits Ulam-Hyers stability in discrete settings, but the system is not Lyapunov stable for a > 0.

From Theorem 3.2 and Remark 3.2 above, one can observe that Ulam-Hyers stability is a more general concept in the stability theory of dynamical systems.

4. Ulam-Hyers stability of linear FDEs

In this section, we first consider the simplest form of a linear FDE,

$$\mathscr{D}_{0,t}^{\alpha}u(t) + \lambda u(t) = g(t), \quad t \in J = [0,T), \quad 0 < T \le +\infty,$$
 (4.1)

where $\lambda \in \mathbb{R}$, $u: J \to \mathbb{R}$, $n-1 < \alpha \leq n$, $n \in \mathbb{N}$, $\mathcal{D}_{0,t}^{\alpha}$ is the Caputo or Riemann-Liouville derivative of order α , and $g: J \to \mathbb{R}$ is a given function. The following results addressing the Ulam-Hyers stability was established by Wang and Xu [79,80] using a Laplace transform.

Theorem 4.1. [79] Given $\varepsilon > 0$, if a function $v: J \to \mathbb{R}$ satisfies the inequality

$$\left|\mathscr{D}_{0,t}^{\alpha}v(t) + \lambda v(t) - g(t)\right| \le \varepsilon, \quad \forall t \in J.$$
 (4.2)

Then, there exists a solution $u: J \to \mathbb{R}$ of the FDE (4.1) such that

$$|v(t) - u(t)| \le \varepsilon t^{\alpha} E_{\alpha,\alpha+1}(|\lambda|t^{\alpha}), \quad \forall t \in J.$$
 (4.3)

Corollary 4.1. If $T < +\infty$, then the linear FDE (4.1) is Ulam-Hyers stable with an Ulam-Hyers constant $L = T^{\alpha}E_{\alpha,\alpha+1}(|\lambda|T^{\alpha})$. However, if $T = +\infty$, no conclusion can be drawn from Theorem 4.1.

Theorem 4.2. [80] Consider the linear FDE (4.1) and let $G: J \to \mathbb{R}_+$ be a given function. Then, if a function $v: J \to \mathbb{R}$ satisfies the inequality

$$\left| \mathcal{D}_{0,t}^{\alpha} v(t) + \lambda v(t) - g(t) \right| \le G(t), \quad \forall t \in J, \tag{4.4}$$

there exists a solution $u: J \to \mathbb{R}$ of the linear FDE (4.1) such that

$$|v(t) - u(t)| \le t^{\alpha} G(t) E_{\alpha, \alpha + 1}(|\lambda| t^{\alpha}), \quad \forall t \in J.$$
(4.5)

Corollary 4.2. If $T < +\infty$, then the linear FDE (4.1) is Ulam-Hyers-Rassias stable with an Ulam-Hyers-Rassias constant $L = T^{\alpha}E_{\alpha,\alpha+1}(|\lambda|T^{\alpha})$. In contrast, if $T=+\infty$, Theorem 4.2 yields no conclusion about its Ulam-Hyers-Rassias stability.

Wang and Li [76] established the Ulam-Hyers stability of the linear FDE for $\lambda \geq 0$, by using the Laplace transform and evaluated the simplified value of the Ulam-Hyers constant. The result is given in next theorem.

Theorem 4.3. [76] Consider the linear FDE (4.1) with $\lambda \geq 0$. Given $\varepsilon > 0$, if a function $v: J \to \mathbb{R}$ satisfies the inequality

$$\left| \mathcal{D}_{0,t}^{\alpha} v(t) + \lambda v(t) - g(t) \right| \le \varepsilon, \quad \forall t \in J.$$
(4.6)

Then, there exists a solution
$$u: J \to \mathbb{R}$$
 of the FDE (4.1) such that
$$|v(t) - u(t)| \leq \frac{t^{\alpha}}{\Gamma(\alpha + 1)} \varepsilon, \quad \forall t \in J. \tag{4.7}$$

Corollary 4.3. If $T < +\infty$, then the linear FDE (4.1) with $\lambda \geq 0$ is Ulam-Hyers stable with an Ulam-Hyers constant $L = T^{\alpha}/\Gamma(\alpha + 1)$.

In the same paper, the authors also analysed the Ulam-Hyers stability of the linear FDE in a Banach space (X, ||.||) with Caputo derivative of order $0 < \alpha \le 1$:

$$_{C}D_{0,t}^{\alpha}U(t) + AU(t) = H(t), \quad t \in J, \ 0 < \alpha \le 1,$$
 (4.8)

where $H: J \to X$ is a continuous function and $-A: D(A) \subset X \to X$ be the generator of a C_0 -semigroup $\{S(t), t \geq 0\}$, written as $S(t) = e^{At}$ on the Banach space X. Denote $M = \sup_{t>0} ||S(t)||$. The result is then

Theorem 4.4. [76] Given $\varepsilon > 0$, if a function $V: J \to \mathbb{R}$ satisfies the inequality

$$\left| {}_{C}\mathrm{D}_{0\,t}^{\alpha}V(t) + AV(t) - H(t) \right| \le \varepsilon, \quad t \in J, \tag{4.9}$$

then there exists a solution $U: J \to \mathbb{R}$ of the FDE (4.8) such that

$$|V(t) - U(t)| \le \frac{Mt^{\alpha}}{\Gamma(\alpha + 1)} \varepsilon, \quad \forall t \in J.$$
 (4.10)

Shen and Chen [62] discussed the Ulam stability of the generalised linear FDE with constant coefficients involving a Riemann-Liouville fractional derivative by the Laplace transform method and evaluated the values of the Ulam-Hyers constant in the integral form. First they considered the following linear FDE

$$_{RL}D_{0,t}^{\alpha}u(t) - \lambda_{RL}D_{0,t}^{\beta}u(t) - \theta u(t) = g(t), \quad t \in J,$$
 (4.11)

where $\lambda, \theta \in \mathbb{R}$, $n-1 < \alpha \le n$ $(n \in \mathbb{N})$, and $0 < \beta < \alpha$, and established the result

Theorem 4.5. [62] Let g(t) be a given function such that

$$\int_0^t (t-\tau)^{\alpha-1} G_{\alpha,\beta;\lambda,\theta}(t-\tau) g(\tau) d\tau$$

exists for any $t \in J$. Suppose that $\varphi : J \to \mathbb{R}_+$ is a function such that the integral

$$\int_0^t (t-\tau)^{\alpha-1} \varphi(\tau) \left| G_{\alpha,\beta;\lambda,\theta}(t-\tau) \right| \, d\tau$$

exists for any $t \in J$. If a function $v: J \to \mathbb{R}$ satisfies the inequality

$$\left|_{RL} \mathcal{D}_{0,t}^{\alpha} v(t) - \lambda_{RL} \mathcal{D}_{0,t}^{\beta} v(t) - \theta v(t) - g(t) \right| \leq \varphi(t), \quad \forall t \in J.$$

Then, there exists a solution $u: J \to \mathbb{R}$ of the linear FDE (4.11) such that

$$|v(t) - u(t)| \le \int_0^t (t - \tau)^{\alpha - 1} \varphi(\tau) |G_{\alpha, \beta; \lambda, \theta}(t - \tau)| d\tau, \quad \forall t \in J,$$

$$(4.12)$$

provided that the series

$$G_{\alpha,\beta;\lambda,\theta}(t) = \sum_{k=0}^{\infty} \frac{\theta^k}{k!} t^{\alpha k} {}_{1} \Psi_1 \begin{bmatrix} (k+1,1) \\ (\alpha k + \alpha, \alpha - \beta) \end{bmatrix} \lambda t^{\alpha-\beta}$$

$$(4.13)$$

is convergent. Here, $_1\Psi_1$ is the hypergeometric function [63].

For $\theta = 0$, and $\lambda \leq 0$, in the above FDE (4.11), using Lemma 2.1 and Theorem 4.5, the Ulam-Hyers stability was studied for the linear FDE

$$_{RL}D_{0,t}^{\alpha}u(t) - \lambda_{RL}D_{0,t}^{\beta}u(t) = g(t), \quad t \in J,$$
 (4.14)

as a corollary, which is given below.

Corollary 4.4. Let $\lambda \leq 0$, $n-1 < \alpha \leq n$ $(n \in \mathbb{N})$, $0 < \beta < \alpha$. Given $\varepsilon > 0$, if a function $v : J \to \mathbb{R}$ satisfies the inequality

$$\left|_{RL} \mathcal{D}_{0,t}^{\alpha} v(t) - \lambda_{RL} \mathcal{D}_{0,t}^{\beta} v(t) - g(t) \right| \leq \varepsilon, \quad \forall t \in J.$$

Then there exists a solution $u: J \to \mathbb{R}$ of the linear FDE (4.14) such that

$$|v(t) - u(t)| \le \frac{t^{\alpha}}{\Gamma(\alpha + 1)} \varepsilon, \quad \forall t \in J$$

Here, we note that if $T < +\infty$, then the linear FDE (4.14) is Ulam-Hyers stable with an Ulam-Hyers constant $L = T^{\alpha}/\Gamma(\alpha + 1)$.

Finally, they presented the Ulam-Hyers-Rassias stability result for the following generalised linear FDE

$${}_{RL}D_{0,t}^{\alpha}u(t) - \lambda_{RL}D_{0,t}^{\beta}u(t) - \sum_{k=0}^{m-2} A_{kRL}D_{0,t}^{\alpha_k}u(t) = g(t), \quad t \in J,$$
(4.15)

where $n-1 < \alpha \le n \ (n \in \mathbb{N}), \ \alpha > \beta > \alpha_{m+2} > \ldots > \alpha_0 = 0, \ \lambda, A_k \in \mathbb{R}, \ k = 0, 1, \ldots, m-2 \ (m \in \mathbb{N} \setminus \{1, 2\}) \text{ and } g : J \to \mathbb{R} \text{ is a given function. The result is}$

Theorem 4.6. [62] Let g(t) be a given function such that

$$\int_0^t (t-\tau)^{\alpha-1} G_{\alpha_1,\dots,\alpha_{m-2},\beta,\alpha;\lambda}(t-\tau)g(\tau) d\tau$$

exists for any $t \in J$. Suppose $\varphi: J \to \mathbb{R}_+$ is a function such that the integral

$$\int_0^t (t-\tau)^{\alpha-1} \varphi(\tau) |G_{\alpha_1,\dots,\alpha_{m-2},\beta,\alpha;\lambda}(t-\tau)| d\tau$$

exists for any $t \in J$. If a function $v: J \to \mathbb{R}$ satisfies the inequality

$$\left| \underset{RL}{\operatorname{D}}_{0,t}^{\alpha} v(t) - \lambda_{RL} \operatorname{D}_{0,t}^{\beta} v(t) - \sum_{k=0}^{m-2} A_{kRL} \operatorname{D}_{0,t}^{\alpha_k} v(t) - g(t) \right| \leq \varphi(t), \quad \forall t \in J.$$

Then, there exists a solution $u: J \to \mathbb{R}$ of the generalised linear FDE (4.15) such that

$$|v(t) - u(t)| \le \int_0^t (t - \tau)^{\alpha - 1} \varphi(\tau) |G_{\alpha_1, \dots, \alpha_{m-2}, \beta, \alpha; \lambda}(t - \tau)| d\tau, \quad \forall t \in J,$$
 (4.16)

provided that the series

$$G_{\alpha_{1},\dots,\alpha_{m-2},\beta,\alpha;\lambda}(t) = \sum_{l=0}^{\infty} \left(\sum_{i_{0}+i_{1}+\dots+i_{m-2}=l} \right) \frac{1}{i_{0}! i_{1}! \cdots i_{m-2}!} \left[\prod_{p=0}^{m-2} (A_{p})^{i_{p}} \right] \times t^{(\alpha-\beta)l+\sum_{p=0}^{m-2} (\beta-\alpha_{p})i_{p}} \times t^{(\alpha-\beta)l+\sum_{p=0}^{m-2} (\beta-\alpha_{p})i_{p}} \times 1 \left[((\alpha-\beta)l+\sum_{p=0}^{m-2} (\beta-\alpha_{p})i_{p}, \alpha-\beta) \right] \lambda t^{\alpha-\beta}$$

$$(4.17)$$

is convergent.

Liu and Li [41] considered the Ulam-Hyers stability of linear FDEs with variable coefficients involving both Riemann-Liouville and Caputo fractional derivatives on a bounded interval I = [0, a],

$$\mathcal{D}_{0,t}^{\alpha}u(t) + q(t)u(t) = r(t), \quad t \in I, \tag{4.18}$$

where $n-1 < \alpha \le n$, $(n \in \mathbb{N})$ and $\mathcal{D}_{0,t}^{\alpha}$ is the Caputo or Riemann-Liouville derivative of order α ; moreover, r(t), q(t) are given continuous functions on I = [0, a]. By using Grönwall's inequality, they established the Ulam-Hyers stability result given below.

Theorem 4.7. [41] Assume there exists a constant K > 0 such that

$$|(t-\tau)^{\alpha-1}q(\tau)| \le K, \ \forall \tau \in [0,t],$$
 (4.19)

for each 0 < t < a. Given $\varepsilon > 0$, if a function $v : I \to \mathbb{R}$ satisfies the inequality

$$|\mathscr{D}_{0,t}^{\alpha}v(t)+q(t)v(t)-r(t)|\leq\varepsilon,\quad\forall\,t\in I.$$

Then, there exists a constant L > 0 and a solution $u : I \to \mathbb{R}$ of the FDE (4.18) such that

$$|v(t) - u(t)| \le L\varepsilon, \forall t \in I,$$
 (4.20)

where

$$L = \frac{a^{\alpha}}{\Gamma(\alpha + 1)} \left[1 + \frac{aK}{\Gamma(\alpha)} \exp\left(\frac{aK}{\Gamma(\alpha)}\right) \right]$$

is the Ulam-Hyers constant.

The Ulam-Hyers stability of a general linear functional equation on a Banach space was studied by Takagi et al. [66]. They also derived an expression for the best Ulam-Hyers constant. Before we discuss their result, we briefly recall some definitions concerning the Ulam-Hyers stability of linear functional equations.

Let $(X, ||.||_X)$, $(Y, ||.||_Y)$ be the normed linear spaces and consider a linear map $T: X \to Y$.

Definition 4.1. [66] We say that the linear map T has Ulam-Hyers stability, if there exists a constant L > 0 with the following property: For any $h \in T(X)$, $\varepsilon > 0$ and $g \in X$ satisfying $||Tg - h||_Y \le \varepsilon$, there exists a $g_0 \in X$ with $Tg_0 = h$, such that $||g - g_0||_X \le L \cdot \varepsilon$.

Now, let T be the bounded linear map and $\mathcal{N}(T)$, $\mathcal{R}(T)$ the kernel and range of T respectively. Define an induced one-to-one map $\tilde{T}: X/\mathcal{N}(T) \to Y$, where $X/\mathcal{N}(T)$ is a quotient space as:

$$\tilde{T}(g + \mathcal{N}(T)) = Tg, \quad g \in X.$$

Let $\tilde{T}^{-1}: \mathcal{R}(T) \to X/\mathcal{N}(T)$ be the inverse of \tilde{T} .

Theorem 4.8. [66] Let X, Y be Banach spaces and $T: X \to Y$ be a bounded linear map. Then, the following statements are equivalent:

- (i) T is Ulam-Hyers stable,
- (ii) $\mathcal{R}(T)$ is bounded,
- (iii) \tilde{T}^{-1} is bounded.

Moreover, if one of the conditions (i), (ii), or (iii) is true. Then, the best Ulam-Hyers constant is given $L_M = ||\tilde{T}^{-1}||$.

Based on Theorem 4.8, we observe that if a linear functional equation involving bounded linear operators is Ulam-Hyers stable, then the best Ulam-Hyers constant for this equation exists. Thus, by choosing an appropriate functional space and a norm, one can prove the existence of the best Ulam-Hyers constant for the linear FDE.

5. Ulam-Hyers stability of non-linear FDEs

In this section, we present the Ulam-Hyers stability results for non-linear FDEs. Wang et al. [78] addressed the Ulam-Hyers stability of non-linear FDEs involving a Caputo fractional derivative of order $\alpha \in (0,1)$ by using the Henry-Grönwall inequality. They also analysed the dependence of data for non-linear FDEs in the case $1 < \alpha < 2$. In essence, they studied the Ulam stability of the following FDE:

$$_{C}D_{a,t}^{\alpha}u(t) = f(t, u(t)), \quad t \in [a, b), a < b \le +\infty,$$
 (5.1)

where $0 < \alpha < 1$, $f : [a, b) \times X \to X$, and $(X, \|.\|_X)$ is a Banach space. Under the following assumptions on f:

- (A1) $f \in C([a,b) \times X,X);$
- (A2) There exists a constant $l_f > 0$ such that

$$||f(t, u_1) - f(t, u_2)||_X \le l_f ||u_1 - u_2||_X,$$

for each $t \in [a, b)$ and for all $u_1, u_2 \in X$;

(A3) Let $\varphi \in C([a,b),\mathbb{R}_+)$ be a non-decreasing function and there exits a constant $C_{\varphi} > 0$ such that

$$\frac{1}{\Gamma(\alpha)} \int_{a}^{t} (t - \tau)^{\alpha - 1} \varphi(\tau) \, d\tau \le C_{\varphi} \varphi(t),$$

for each $t \in [a, b)$.

Theorem 5.1. [78] Let assumptions (A1), (A2), and (A3) hold. If a function $v:[a,b) \to X$ satisfies the inequality

$$\|_C \mathcal{D}_{a,t}^{\alpha} v(t) - f(t, v(t))\|_X \le \varphi(t), \quad \forall t \in [a, b),$$

then there exists a solution $u:[a,b)\to X$ of the FDE (5.1) such that

$$||v(t) - u(t)||_X \le C_{\varphi} \varphi(t) E_{\alpha}(l_f(t - a)^{\alpha}), \quad \forall t \in [a, b).$$

$$(5.2)$$

Corollary 5.1. If $b < +\infty$, then the FDE (5.1) is Ulam-Hyers-Rassias stable with $L = C_{\varphi}E_{\alpha}((b-a)^{\alpha}l_f)$. However, if $b = +\infty$, no conclusion can be drawn from Theorem 5.1.

They also established the following Ulam-Hyers stability results for $0 < \alpha < 1$.

Theorem 5.2. [78] Assume (A1) and (A2) hold. Let $\varepsilon > 0$ be a given number and suppose a function $v : [a, b) \to X$ satisfies the inequality

$$\|_{C} \mathcal{D}_{a,t}^{\alpha} v(t) - f(t, v(t))\|_{X} \le \varepsilon, \quad \forall t \in [a, b),$$

then there exists a solution $u:[a,b)\to X$ of the FDE (5.1) such that

$$\|v(t) - u(t)\|_{X} \le \frac{\varepsilon(t-a)^{\alpha}}{\Gamma(\alpha+1)} E_{\alpha}(l_f(t-a)^{\alpha}), \quad \forall t \in [a,b).$$
 (5.3)

Corollary 5.2. If $b < +\infty$, then the FDE (5.1) is Ulam-Hyers stable with

$$L = \frac{(b-a)^{\alpha}}{\Gamma(\alpha+1)} E_{\alpha}((b-a)^{\alpha} l_f).$$

However, if $b = +\infty$, no conclusion can be drawn from Theorem 5.2.

In [77] Wang et al. also investigated the Ulam stability of the same non-linear FDE (5.1) with $X = \mathbb{R}$ on a closed and bounded interval by using the Banach fixed point theorem. Further, they improved and simplified the Ulam-Hyers constant. Their results are given below.

Theorem 5.3. [77] For $X = \mathbb{R}$, let assumptions (A1) and (A2) hold over the finite interval [a, b]. Suppose that $\varphi : [a, b] \to \mathbb{R}_+$ is a continuous function with

$$\frac{1}{\Gamma(\alpha)} \int_{a}^{t} (t - \tau)^{\alpha - 1} \varphi(\tau) \, d\tau \le K \varphi(t), \tag{5.4}$$

for each $t \in [a, b]$ and for some K > 0. Let the constants l_f and K satisfy $Kl_f < 1$. If a continuously differentiable function $v : [a, b] \to \mathbb{R}$ satisfies the inequality

$$|_{C} D_{a,t}^{\alpha} v(t) - f(t,v(t))| \le \varphi(t), \quad \forall t \in [a,b],$$

then there exists a unique solution $u:[a,b]\to\mathbb{R}$ of the FDE (5.1) such that

$$|v(t) - u(t)| \le \frac{K}{1 - Kl_f} \varphi(t), \quad \forall t \in [a, b].$$
 (5.5)

In other words, the FDE (5.1) is Ulam-Hyers-Rassias stable.

Here, we notice that the solution u of the FDE (5.1) satisfying (5.5) is unique. In the same paper they also established the Ulam-Hyers stability result given below.

Theorem 5.4. [77] For $X = \mathbb{R}$, let assumptions (A1) and (A2) hold on a finite interval [a, a+h], h > 0. Let the constant l_f satisfy $h^{\alpha}l_f/\Gamma(\alpha+1) < 1$. Given $\varepsilon > 0$, if a continuously differentiable function $v : [a, a+h] \to \mathbb{R}$ satisfies the inequality

$$|{}_{C}\mathrm{D}_{a,t}^{\alpha}v(t) - f(t,v(t))| \le \varepsilon, \quad \forall t \in [a,a+h],$$

then there exists a unique solution $u:[a,a+h]\to\mathbb{R}$ of the FDE (5.1) such that

$$|v(t) - u(t)| \le \frac{h^{\alpha}}{\Gamma(\alpha + 1) - h^{\alpha} l_f} \varepsilon, \quad \forall t \in [a, a + h].$$
 (5.6)

In other words, the FDE (5.1) is Ulam-Hyers stable.

El-Hady and Öğrekci [22] also considered the same non-linear FDE (5.1) with $X = \mathbb{R}$ on a closed and bounded interval [0, r]. They removed the assumption (5.4) on the function φ and studied the Ulam-Hyers-Rassias stability result by using the Banach fixed point theorem on a generalised metric space. Further, they derived the Ulam-Hyers constant with more flexibility on the parameter than in [77]. The corresponding results are given as follows.

Theorem 5.5. [22] For $X = \mathbb{R}$, assume conditions (A1) and (A2) hold on the finite interval [0, r], r > 0. Suppose that $\varphi : [0, r] \to \mathbb{R}_+$ is a continuous and non-decreasing function; if a continuously differentiable function $v : [0, r] \to \mathbb{R}$ satisfies the inequality

$$|{}_{C}\mathrm{D}_{0,t}^{\alpha}v(t)-f(t,v(t))|{\leq \varphi(t)}, \quad \forall \, t \in [0,r],$$

then there exists a unique solution $u:[0,r]\to\mathbb{R}$ of the FDE (5.1) such that

$$|v(t) - u(t)| \le \frac{c_2}{c_1 \Gamma(\alpha) - c_2 l_f} \varphi(t), \quad \forall t \in [a, b], \tag{5.7}$$

where c_1 and c_2 are arbitrary positive constants such that the inequalities

$$\frac{c_2 l_f}{c_1 \Gamma(\alpha)} < 1, \quad \max\{1, 2^{1-\alpha}\} \left(\frac{\alpha + 1}{\alpha}\right) \le c_2 e^{c_1 t} \tag{5.8}$$

holds for each $t \in [0, r]$ In other words, the FDE (5.1) is Ulam-Hyers-Rassias stable.

Corollary 5.3. For $X = \mathbb{R}$, let the assumptions (A1) and (A2) hold on the finite interval [0, r], r > 0. Then, the FDE (5.1) is Ulam-Hyers stable with $L = c_2/[c_1\Gamma(\alpha) - c_2l_f]$, provided that the inequalities (5.8) hold.

Hristove and Abbas [30] investigated the existence of the solution and the Ulamtype stability of an initial value problem (IVP) for non-linear FDEs involving a generalised proportional fractional derivative of Riemann-Liouville fractional type on a closed and bounded interval [a, b]. They established the results by using the Henry-Gröwall inequality and applied them to a fractional generalisation of a biological population model as an application. They considered the following non-linear FDE (IVP):

$$RL D_{a,t}^{\alpha,\rho} u(t) = \lambda u(t) + f(t, u(t)), \quad t \in [a, b],
\mathcal{I}_{a,t}^{1-\alpha,\rho} u(a) = u_0,$$
(5.9)

where $0 < \alpha < 1$, $0 < \rho \le 1$, and λ, u_0 are real constants, and $f : [a, b] \times \mathbb{R} \to \mathbb{R}$ is a given continuous function. The solution of the above initial value problem (5.9) exists on a Banach space

$$C_{1-\alpha,\rho}([a,b],\mathbb{R}) = \left\{ x : (a,b] \to \mathbb{R} \mid x \in C((a,b],\mathbb{R}), \right.$$

$$\lim_{t \to a^{+}} e^{\frac{1-\rho}{\rho}(t-a)} (t-a)^{1-\alpha} x(t) < +\infty \right\},$$

with a norm $||x||_{C_{1-\alpha,\rho}} = \max_{t \in [a,b]} \left| \exp\left(\frac{1-\rho}{\rho}(t-a)\right)(t-a)^{1-\alpha}x(t) \right|$. The solution $u \in C_{1-\alpha,\rho}([a,b],\mathbb{R})$ satisfies the following integral equation

$$u(t) = u_0 e^{\frac{\rho - 1}{\rho}(t - a)} E_{\alpha,\alpha} \left(\lambda \left(\frac{t - a}{\rho} \right)^{\alpha} \right) \left(\frac{t - a}{\rho} \right)^{\alpha - 1}$$

$$+ \frac{1}{\rho^{\alpha} \Gamma(\alpha)} \int_a^t (t - \tau)^{\alpha - 1} \exp\left(\frac{\rho - 1}{\rho} (t - \tau) \right) E_{\alpha,\alpha} \left(\lambda \left(\frac{t - \tau}{\rho} \right)^{\alpha} \right) f(\tau, u(\tau)) d\tau,$$

$$(5.10)$$

for t > a.

Theorem 5.6. [30] Let $\lambda \in \mathbb{R}$ and assume the conditions (A1) and (A2) hold with $X = \mathbb{R}$. Let $\varepsilon > 0$ be an arbitrary given number and suppose a function $v \in C_{1-\alpha,\rho}([a,b])$ satisfies the inequality

$$\left| P_{RL} D_{a,t}^{\alpha,\rho} v(t) - \lambda v(t) - f(t,v(t)) \right| \le \varepsilon, \quad \forall t \in [a,b],$$

then there exists a solution $u \in C_{1-\alpha,\rho}([a,b])$ of the IVP (5.9) such that

$$|v(t) - u(t)| \le L\varepsilon, \quad \forall t \in [a, b],$$
 (5.11)

where

$$L = \left[\frac{1}{|\lambda| \Gamma(\alpha)} \left| E_{\alpha} \left(\lambda \left(\frac{b-a}{\rho} \right)^{\alpha} \right) - 1 \right| + e_{1} \left(\frac{b-a}{\rho} \right)^{\alpha-1} \right] E_{\alpha} \left(e_{2} l_{f} \left(\frac{b-a}{\rho} \right)^{\alpha} \right),$$

$$e_{1} = \max_{t \in [a,b]} E_{\alpha,\alpha} \left(\lambda \left(\frac{t-a}{\rho} \right)^{\alpha} \right), \quad e_{2} = \max_{t,s \in [a,b]; t \geq s} E_{\alpha,\alpha} \left(\lambda \left(\frac{t-s}{\rho} \right)^{\alpha} \right). \tag{5.12}$$

In other words, the FDE (5.9) is Ulam-Hyers stable.

For $\lambda < 0$, they proved the Ulam-Hyers-Rassias stability for the FDE (5.9) and estimated the simplified value of the Ulam-Hyers constant as follows.

Theorem 5.7. [30] Let $\lambda < 0$ and assume the assumptions (A1), (A2), and (A3) hold on bounded interval [a,b] with $X = \mathbb{R}$. If a function $v \in C_{1-\alpha,\rho}([a,b])$ satisfies the inequality

$$\left| {_{RL}} \mathcal{D}_{a,t}^{\alpha,\rho} v(t) - \lambda v(t) - f(t,v(t)) \right| \le \varphi(t), \quad \forall \, t \in [a,b],$$

then there exists a solution $u \in C_{1-\alpha,\rho}([a,b])$ of the IVP (5.9) such that

$$|v(t) - u(t)| \le L\varphi(t), \quad \forall t \in [a, b], \tag{5.13}$$

where

$$L = \left[\frac{\varphi(b)C_{\varphi}}{\rho^{\alpha}\Gamma(\alpha)} e_2 + e_1\varphi(a) \left(\frac{b-a}{\rho} \right)^{\alpha-1} \right] E_{\alpha} \left(e_2 l_f \left(\frac{b-a}{\rho} \right)^{\alpha} \right), \tag{5.14}$$

and e_1, e_2 are defined in Theorem 5.6.

In [18] Cuong presented the Ulam-Hyers stability analysis of multi-order FDEs involving the Riemann-Liouville derivative. They established the Ulam stability with respect to a $\|.\|_{C_{\gamma}}$ -norm on a $C_{\gamma}([0,T],\mathbb{R}^d)$ space $(0 \leq \gamma < 1, d \in \mathbb{N})$ followed by a Bielecki type norm. The $C_{\gamma}([0,T],\mathbb{R}^d)$ space is defined as

$$C_{\gamma}([0,T], \mathbb{R}^d) = \{ \mathbf{u} \in C((0,T], \mathbb{R}^d) \mid \sup_{t \in [0,T]} ||t^{\gamma} \mathbf{u}(t)||_{\mathbb{R}^d} < \infty \}, \quad 0 \le \gamma < 1,$$

with a norm

$$\|\mathbf{u}\|_{C_{\gamma}} = \sup_{t \in [0,T]} \|t^{\gamma}\mathbf{u}(t)\|_{\mathbb{R}^d}.$$
 (5.15)

where $\|\cdot\|_{\mathbb{R}^d}$ is a norm on \mathbb{R}^d . Basically, they studied the Ulam-type stability for the following initial value FDE (IVP):

$$_{RL}\mathcal{D}_{0,t}^{\tilde{\alpha}}\mathbf{u}(t) = \mathbf{f}(t,\mathbf{u}(t)), \quad t \in (0,T],$$

$$(5.16)$$

with initial condition

$$\lim_{t\to 0^+} \operatorname{diag}[t^{1-\alpha_1},\ldots,t^{1-\alpha_d}]\mathbf{u}(t) = \mathbf{u}_0,$$

where $\mathbf{u}(t) = (u_1(t), \dots, u_d(t)) \in \mathbb{R}^d$, $\tilde{\alpha} = (\alpha_1, \dots, \alpha_d)$, $0 \le \alpha_i < 1$, $i = 1, \dots, d$, and $\mathbf{f} : [0, T] \times \mathbb{R}^d \to \mathbb{R}^d$. The Riemann-Liouville multi-order fractional derivative $R_L D_{0,t}^{\tilde{\alpha}} \mathbf{u}(t)$ is defined by $R_L D_{0,t}^{\tilde{\alpha}} \mathbf{u}(t) = (R_L D_{0,t}^{\alpha_1} u_1(t), \dots, R_L D_{0,t}^{\alpha_d} u_d(t))$.

Lemma 5.1. [18] Assume that **f** is continuous on $[0,T] \times \mathbb{R}^d$. Then, a function $\mathbf{u} \in C((0,T],\mathbb{R}^d)$ is a solution of the IVP (5.16) if and only if it is a solution of the Volterra integral equation

$$\mathbf{u}(t) = \operatorname{diag}[t^{\alpha_1 - 1}, \dots, t^{\alpha_d - 1}]\mathbf{u}_0 + {}_{RL}\mathbf{I}_{0,t}^{\tilde{\alpha}}\mathbf{f}(t, \mathbf{u}(t)), \quad \forall t \in (0, T],$$
(5.17)

where $_{RL}I_{0,t}^{\tilde{\alpha}}$ is a multi-order fractional integral operator.

Since the integral equation (5.17) is an equivalent form of the FDE (5.16). Cuong studied the Ulam-Hyers stability of the equivalent integral equation (5.17) instead of the considered FDE (5.16). The result is given below.

Theorem 5.8. [18] Assume \mathbf{f} is continuous and Lipschitz continuous with respect to the second variable with a Lipschitz constant $L_{\mathbf{f}} > 0$. Let $\varepsilon > 0$ be an arbitrary given number and suppose a function $\mathbf{v} \in C_{1-\alpha_0}([0,T],\mathbb{R}^d)$ satisfies the inequality

$$\sup_{t \in [0,T]} \left\| t^{1-\alpha_0} \left(\mathbf{v}(t) - \operatorname{diag}[t^{\alpha_1 - 1}, \dots, t^{\alpha_d - 1}] \mathbf{v}_0 - {}_{RL} \mathbf{I}_{0,t}^{\tilde{\alpha}} \mathbf{f}(t, \mathbf{v}(t)) \right) \right\|_{\mathbb{R}^d} \le \varepsilon, \quad (5.18)$$

then there exists a solution $\mathbf{u} \in C_{1-\alpha_0}([0,T],\mathbb{R}^d)$ of the FDE (5.16) such that

$$\sup_{t \in [0,T]} \| t^{1-\alpha_0} \left(\mathbf{v}(t) - \mathbf{u}(t) \right) \|_{\mathbb{R}^d} \le \frac{e^{\theta T}}{1-\sigma}, \tag{5.19}$$

where $\alpha_0 = \max\{\alpha_1, \dots \alpha_d\}$ and $\theta > 0$ is chosen large enough such that

$$\sigma = \max_{i \in \{1, \dots, d\}} \left\{ L_{\mathbf{f}} \left(\frac{2^{1 - \alpha_i} T^{\alpha_i - \alpha_0} \Gamma(\alpha_0)}{\theta^{\alpha_0} \Gamma(\alpha_i)} + \frac{2^{1 - \alpha_0}}{\theta^{\alpha_i}} \right) \right\} < 1.$$

In the other words, the multi-order FDE (5.16) is Ulam-Hyers stable with respect to the $\|.\|_{C_{1-\alpha_0}}$ norm.

6. Fractional-order delay differential equations

A delay differential equation (DDE) is a general class of differential equation in dynamical systems, occurring naturally when modelling real-world problems. For instance, any feedback control system inherently includes time delays, as sensing information and responding to it takes finite time. Integer-order DDEs have a welldeveloped theory regarding the existence and stability of solutions [23,39,45,50,59]. In recent decades, fractional delay differential equations (FDDEs) have received considerable attention due to their applications in dynamical systems and control. For results on the existence and stability of solutions to FDDEs, see [17,69].

Refaai et al. [57] studied the Ulam-type stability of FDDEs involving a Riemann-Liouville fractional derivative in a closed and bounded interval by using the Banach fix point theorem followed by Henry-Gröwall inequality. However, according to our analysis their assumptions and derivations are not correct. For a detailed proof see [57]. Develi and Duman [20] studied the existence of solutions and the Ulam-Hyers stability of FDDEs involving a Caputo fractional derivative by using the Banach fixed point theorem on a Banach space $C([-\theta, b], \mathbb{R})$ endowed with the following Bielecki norm.

$$||v||_B = \max_{t \in [-\theta, b]} |v(t)| e^{-\gamma t}, \quad \gamma > 0.$$
 (6.1)

Concretely, they consider the following delay system:

$${}_{C}\mathrm{D}_{0,t}^{\alpha}u(t) = h(t, u(t), u(k(t))), \ t \in [0, b], \ b > 0,$$

$$u(t) = \zeta(t), \quad t \in [-\theta, 0],$$
(6.2)

$$u(t) = \zeta(t), \quad t \in [-\theta, 0], \tag{6.3}$$

where $0 < \alpha \leq 1, 0 < \theta < \infty, h \in C([0,b] \times \mathbb{R}^2, \mathbb{R}), \zeta \in C([-\theta,0], \mathbb{R}),$ and $k \in C([0,b],[-\theta,b])$ with $k(t) \leq t$. Under the following assumptions:

- **(A4)** $h \in C([0,b] \times \mathbb{R}^2, \mathbb{R}), k \in C([0,b], [-\theta,b]) \text{ with } k(t) \leq t \text{ on } [0,b].$
- (A5) There exists a constant $\omega > 0$ such that

$$|h(t, x_1, y_1) - h(t, x_2, y_2)| \le \omega(|x_1 - x_2| + |y_1 - y_2|)$$

for all $x_i, y_i \in \mathbb{R}$ (i = 1, 2) and $t \in [0, b]$, the following result was established.

Theorem 6.1. [20] Suppose the assumptions (A4) and (A5) hold. Let $\varepsilon > 0$ be an arbitrary given number and suppose a function $v \in C([-\theta, b], \mathbb{R})$ satisfies the inequality

$$\left|{}_{C}\mathrm{D}_{0,t}^{\alpha}v(t)-h(t,v(t),v(k(t)))\right|\leq\varepsilon,\quad\forall\,t\in[0,b],$$

then there exists a solution $u \in C([-\theta, b], \mathbb{R})$ of the delay system (6.2) such that

$$|v(t) - u(t)| \le \frac{b^{\alpha} e^{(\theta + b)\gamma}}{(1 - \rho)\Gamma(\alpha + 1)} \varepsilon, \quad \forall t \in [-\theta, b], \tag{6.4}$$

where $\gamma > 0$ is an appropriate real number such that $\rho = 2\omega/\gamma^{\alpha} < 1$.

Recently, Benzarouala and Tunc [10] studied the Ulam-type stability of the FDDEs involving a Caputo derivative with n-multiple variable time delays:

$${}_{C}D_{a,t}^{\alpha}u(t) = \sum_{i=1}^{n} B_{i}F_{i}(t, u(t), u(h_{i}(t))), \quad t \in [a, b],$$

$$u(t) = \zeta(t), \quad t \in [a - \theta, a],$$
(6.5)

where $0 < \alpha \le 1$, $\zeta \in C([a - \theta, a], \mathbb{R})$, $F_i \in C([a, b] \times \mathbb{R} \times \mathbb{R}, \mathbb{R})$, $B_i \in \mathbb{R}$ for $i = 1, \ldots, n$, and $h_i \in C([a, b], [a - \theta, b])$ with $h_i(t) \le t$ such that $0 \le h_i(t) \le \theta_i$, $\theta = \max\{\theta_i : i = 1, \ldots, n\}$.

The result was established by the utilisation of the Banach fixed point theorem on a complete metric space X given by

$$X = \{ v \in C([a - \theta, b], \mathbb{R}) : f(t) = \zeta(t), \text{ if } t \in [a - \theta, a] \}$$

endowed with the metric

$$d(v_1, v_2) = \inf\{C > 0 : |v_1(t) - v_2(t)| \le C\varphi(t), \ t \in [a, b]\},\$$

where $\varphi:[a,b]\to\mathbb{R}_+$ is a continuous function. Along with the following assumptions:

(A6) For every $i = 1, \dots, n$, $F_i \in C([a, b] \times \mathbb{R} \times \mathbb{R}, \mathbb{R})$ there exists positive constants ω_i and $\tilde{\omega}_i$ such that

$$|F_i(t, x_1, y_1) - F_i(t, x_2, y_2)| \le (\omega_i |x_1 - x_2| + \tilde{\omega}_i |y_1 - y_2|)$$

for every $t \in [a, b]$ and for all $x_i, y_i \in \mathbb{R}$ (i = 1, 2).

(A7) Let $\varphi \in C([a-\theta,b],\mathbb{R}_+)$ be a non-decreasing function and there exits a constant $L_{\varphi} > 0$ such that

$$\int_{a}^{t} (t-\tau)^{\alpha-1} \varphi(\tau) d\tau \le L_{\varphi} \varphi(t),$$

for each $t \in [a, b]$.

Theorem 6.2. [10] Assume that the assumptions (A6) and (A7) hold. If a function $v \in C^1([a-\theta,b],\mathbb{R})$ satisfies the inequalities

$$\begin{cases} \left| {}_{C}\mathrm{D}_{a,t}^{\alpha}v(t) - \sum_{i=1}^{n}B_{i} \cdot F_{i}(t,v(t),v(h_{i}(t))) \right| \leq \varphi(t), & t \in [a,b], \\ \left| v(t) - \zeta(t) \right| \leq \varphi(t), & t \in [a-\theta,a], \end{cases}$$

then there exists a unique solution $u \in C([a-\theta,b],\mathbb{R})$ of the FDDE (6.5) such that

$$|v(t) - u(t)| \le \frac{L_{\varphi}}{\Gamma(\alpha) - \sum_{i=1}^{n} L_{\varphi}(\omega_i + \tilde{\omega}_i)|B_i|} \varphi(t), \quad \forall t \in [a, b],$$
 (6.6)

provided that $\sum_{i=1}^{n} L_{\varphi}(\omega_i + \tilde{\omega}_i)|B_i| \leq \Gamma(\alpha)$. In other words, the FDDE (6.5) is Ulam-Hyers-Rassias stable.

As a corollary of Theorem 6.2, the following Ulam-Hyers stability results were also established.

Corollary 6.1. Assume the conditions of Theorem 6.2 hold, along with assumption $(\mathbf{A6})$. Let $\varepsilon > 0$ be an arbitrary given number and suppose a function $v \in C^1([a-\theta,b],\mathbb{R})$ satisfies the inequalities

$$\begin{cases} \left| {}_{C}\mathrm{D}_{a,t}^{\alpha}v(t) - \sum_{i=1}^{n}B_{i}\cdot F_{i}(t,v(t),v(h_{i}(t))) \right| \leq \varepsilon, & t \in [a,b], \\ \left| v(t) - \zeta(t) \right| \leq \varepsilon, & t \in [a-\theta,a], \end{cases}$$

then there exists a unique solution $u \in C([a-\theta,b],\mathbb{R})$ of the FDDE (6.5) such that

$$|v(t) - u(t)| \le \frac{(b-a)^{\alpha}}{\Gamma(\alpha+1) - \sum_{i=1}^{n} (b-a)^{\alpha} (\omega_i + \tilde{\omega}_i) |B_i|} \varepsilon, \quad \forall t \in [a, b],$$
 (6.7)

provided that $\sum_{i=1}^{n} (b-a)^{\alpha} (\omega_i + \tilde{\omega}_i) |B_i| < \Gamma(\alpha+1)$. In other words, the FDDE (6.5) is Ulam-Hyers stable.

7. Fractional-order boundary value problem

In the previous section, we discussed the work carried out on the Ulam-type stability of the different classes of FDEs subject to given initial conditions. In this section, some basic Ulam-type stability results of fractional-order boundary value problems (BVPs) will be presented. Applying the fractional-oder model to real-world problems needs a physically interpretable initial/boundary condition which contains $u(0), u'(0), \ldots, u(T), u'(T), \ldots$, etc. There have been multiple studies of the Ulam-type stability of fractional-order BVPs [2,5,11,27,64,73]. Here we present a

few of these results on the Ulam-type stability of the fractional-order BVP involving Caputo fractional derivatives. For the existence of solutions of the different classes of a fractional BVP with a Caputo fractional derivative of order $0 < \alpha < 1$ and $1 < \alpha \le 2$ see the survey paper by Agarwal et al. [1] and the book by Ali et al. [5]. As in the case of an integer-order BVP, the solution of the fractional-order BVP is expressed with the help of the Green's function.

Ali et al. [5] analysed the Ulam-type stability of the following fractional-order BVP:

$${}_{C}\mathrm{D}_{0,t}^{\alpha}u(t) = f(t, u(t)), \quad 1 < \alpha < 2, \ t \in [0, 1],$$

$$\lambda_{1}u(0) + \mu_{1}u(1) = g_{1}(u),$$

$$\lambda_{2}u'(0) + \mu_{2}u'(1) = g_{2}(u),$$

$$(7.1)$$

where g_i $(i = 1, 2) : C([0, 1], \mathbb{R}) \to \mathbb{R}$ are non-local continuous functions, f: $C([0,1]\times\mathbb{R},\mathbb{R})$, and $\lambda_i,\mu_i\in\mathbb{R}$ with $\lambda_i+\mu_i\neq 0$ for i=1,2. The result is obtained by using the Banach fixed point theorem. The solution of the BVP (7.1) is given by

$$u(t) = g(t) + \int_0^1 \mathcal{G}(t, s) f(s, u(s)) ds,$$

$$g(t) = \frac{1}{\lambda_1 + \mu_1} g_1(u) + \frac{1}{\lambda_2 + \mu_2} (t - \mu_1) g_2(u),$$
(7.2)

where

$$g(t) = \frac{1}{\lambda_1 + \mu_1} g_1(u) + \frac{1}{\lambda_2 + \mu_2} (t - \mu_1) g_2(u),$$

and $\mathcal{G}(t,s)$ is the Green's function of the BVP (7.1) given by

$$\mathcal{G}(t,s) = \begin{cases} \frac{(t-s)^{\alpha-1}}{\Gamma(\alpha)} + \frac{\mu_1(1-s)^{\alpha-1}}{(\lambda_1 + \mu_1)\Gamma(\alpha)} \\ + \frac{\mu_2}{\lambda_2 + \mu_2} \left(\frac{\mu_1}{\lambda_1 + \mu_1} - t\right) \frac{(1-s)^{\alpha-2}}{\Gamma(\alpha - 1)}, & 0 \le s \le t \le 1 \\ \frac{\mu_1(1-s)^{\alpha-1}}{(\lambda_1 + \mu_1)\Gamma(\alpha)} + \frac{\mu_2}{\lambda_2 + \mu_2} \left(\frac{\mu_1}{\lambda_1 + \mu_1} - t\right) \frac{(1-s)^{\alpha-2}}{\Gamma(\alpha - 1)}, & 0 \le t \le s \le 1. \end{cases}$$

$$(7.3)$$

To established the result, they assumed the following property of g: (A8) For $u_1, u_2 \in C([0,1], \mathbb{R})$, there exists $c_g \in [0,1)$, such that

$$|g(u_1) - g(u_2)| \le c_g ||u_1 - u_2||_{\infty}; \tag{7.4}$$

where $||u||_{\infty} = \sup_{t \in [0,1]} \{ |u(t)| : u \in C([0,1], \mathbb{R}) \}.$

Theorem 7.1. [5] Assume the assumption (A8) holds and let $f \in C([0,1] \times \mathbb{R}, \mathbb{R})$ be Lipschitz-continuous with respect to the second variable with a Lipschitz constant L_f . Let $\varphi \in C([0,1], \mathbb{R}_+)$ be a non-decreasing function and there exists a constant $\lambda_{\varphi} > 0$ such that $\frac{1}{\Gamma(\alpha)} \int_0^t (t-\tau)^{\alpha-1} \varphi(\tau) ds \leq \lambda_{\varphi} \varphi(t)$ for each $t \in [0,1]$. If a function $v \in C([0,1], \mathbb{R})$ satisfies the inequality

$$\left|{}_{C}\mathrm{D}_{0,t}^{\alpha}v(t)-f(t,v(t))\right|\leq\varphi(t),\quad\forall\,t\in[0,1],$$

then there exists a unique solution $u \in C([0,1],\mathbb{R})$ of the fractional-order BVP (7.1) such that

$$|v(t) - u(t)| \le \frac{\lambda_{\varphi}}{1 - [c_q + \mathcal{G}_0 L_f]} \varphi(t), \quad \forall t \in [0, 1], \tag{7.5}$$

provided that $c_g + \mathcal{G}_0 L_f < 1$, where $\mathcal{G}_0 = \max_{t \in [0,1]} \int_0^1 |\mathcal{G}(t,s)| ds$. In other words, the fractional-order BVP (7.1) is Ulam-Hyers-Rassias stable.

Next, they also established the Ulam-Hyers stability result:

Theorem 7.2. [5] Given the assumption (A8) and $f \in C([0,1] \times \mathbb{R}, \mathbb{R})$ be Lipschitz-continuous with respect to the second variable with a Lipschitz constant L_f . Let $\varepsilon > 0$ be an arbitrary given number and suppose a function $v \in C([0,1],\mathbb{R})$ satisfies the inequality

$$\left| {}_{C}\mathrm{D}_{0,t}^{\alpha}v(t) - f(t,v(t)) \right| \leq \varepsilon, \quad \forall \, t \in [0,1],$$

then there exists a unique solution $u \in C([0,1],\mathbb{R})$ of the fractional-order BVP (7.1) such that

$$|v(t) - u(t)| \le \frac{\mathcal{G}_0}{1 - [c_q + \mathcal{G}_0 L_f]} \varepsilon, \quad \forall t \in [0, 1], \tag{7.6}$$

provided that $c_g + \mathcal{G}_0 L_f < 1$. In other words, the fractional-order BVP (7.1) is Ulam-Hyers stable.

Chen et al. [14] investigated the Ulam-Hyers stability of a class of multi-term non-linear fractional-order BVPs involving a Caputo fractional derivative:

$${}_{C}\mathrm{D}_{0,t}^{\alpha_{1}}u(t) - \xi_{C}\mathrm{D}_{0,t}^{\alpha_{2}}u(t) + f(t, u(t)) = 0, \quad t \in [0, 1],$$

$$u(0) + u(1) = u_{0},$$
 (7.7)

where $0 < \alpha_2 < \alpha_1 \le 1$, $f \in C([0,1] \times \mathbb{R}, \mathbb{R})$, and ξ, u_0 are given constants such that $\xi \ne 2\Gamma(\alpha_1 - \alpha_2 + 1)$. The solution of the fractional BVP (7.7) is given by

$$u(t) = \theta(t) + \int_0^1 \mathcal{H}_1(t,s)u(s) ds - \int_0^1 \mathcal{H}_2(t,s)f(s,u(s)) ds,$$
 (7.8)

where

$$\theta(t) = \left(\frac{\xi t^{\alpha_1 - \alpha_2} - \Gamma(\alpha_1 - \alpha_2 + 1)}{\xi - 2\Gamma(\alpha_1 - \alpha_2 + 1)}\right) u_0$$

as well as

$$\mathcal{H}_{1}(t,s) = \frac{\xi}{\Gamma(\alpha_{1} - \alpha_{2})} \begin{cases} \frac{\xi t^{\alpha_{1} - \alpha_{2}} - \Gamma(\alpha_{1} - \alpha_{2} + 1)}{2\Gamma(\alpha_{1} - \alpha_{2} + 1) - \xi} (1 - s)^{\alpha_{1} - \alpha_{2} - 1} \\ + (t - s)^{\alpha_{1} - \alpha_{2} - 1}, & 0 \le s \le t \le 1 \end{cases}$$

$$\frac{\xi t^{\alpha_{1} - \alpha_{2}} - \Gamma(\alpha_{1} - \alpha_{2} + 1)}{2\Gamma(\alpha_{1} - \alpha_{2} + 1) - \xi} (1 - s)^{\alpha_{1} - \alpha_{2} - 1}, \quad 0 \le t \le s \le 1$$

and

$$\mathcal{H}_{2}(t,s) = \frac{1}{\Gamma(\alpha_{1})} \begin{cases} \frac{\xi t^{\alpha_{1}-\alpha_{2}} - \Gamma(\alpha_{1}-\alpha_{2}+1)}{2\Gamma(\alpha_{1}-\alpha_{2}+1) - \xi} (1-s)^{\alpha_{1}-1} \\ +(t-s)^{\alpha_{1}-1}, & 0 \leq s \leq t \leq 1, \\ \frac{\xi t^{\alpha_{1}-\alpha_{2}} - \Gamma(\alpha_{1}-\alpha_{2}+1)}{2\Gamma(\alpha_{1}-\alpha_{2}+1) - \xi} (1-s)^{\alpha_{1}-1}, & 0 \leq t \leq s \leq 1. \end{cases}$$

By using the Banach fixed point theorem and the Grönwall inequality, they obtained the following result.

Theorem 7.3. [14] Assume $\mathcal{M}_0 = \max_{t \in [0,1]} \int_0^1 |\mathcal{H}_1(t,s)| \, ds < 1 \, and \, \xi \neq \Gamma(\alpha_1 - \alpha_2 + 1).$ Let $f \in C([0,1] \times \mathbb{R}, \mathbb{R})$ be Lipschitz continuous with respect to second variable with a Lipschitz constant L_f . Given any $\varepsilon > 0$, if a function $v \in C([0,1], \mathbb{R})$ satisfies the inequality

$$\left| {}_{C}\mathrm{D}_{0,t}^{\alpha_{1}}v(t) - \xi_{C}\mathrm{D}_{0,t}^{\alpha_{2}}v(t) + f(t,v(t)) \right| \leq \varepsilon, \quad \forall t \in [0,1],$$

then there exists a unique solution $u \in C([0,1],\mathbb{R})$ such that

$$|v(t) - u(t)| \le \frac{M \left(3^{p-1} \exp(3^{p-1}Q_1^p)\right)^{1/p}}{\left(1 - L_0^p \left[\exp\left(3^{p-1}Q_1^p\right) - 1\right]\right)^{1/p}} \varepsilon, \quad \forall t \in [0, 1], \tag{7.9}$$

where

$$L_0 = \frac{|\xi| + \Gamma(\alpha_1 - \alpha_2 + 1)}{|\xi - \Gamma(\alpha_1 - \alpha_2 + 1)|}, \quad M = \frac{1}{\Gamma(\alpha_1 + 1)} (1 + |\xi|L_0)$$

and

$$Q_1 = \frac{|\xi|}{\Gamma(\alpha_1 - \alpha_2)(1 + (\alpha_1 - \alpha_2 - 1)q)^{1/q}} + \frac{k_f}{\Gamma(\alpha_1)(1 + (\alpha_1 - 1)q)^{1/q}}.$$

Moreover, $p, q \in (1, \infty)$ such that 1/p + 1/q = 1 and $\alpha_1 - \alpha_2 + 1/q > 1$.

From the above results, we note that compared to the fractional-order initial value problem (IVP) the Ulam-type stability for the fractional-order BVP is quite complex to analyse. One of the reasons is that the IVP utilises different methods such as integral transform method, fixed point method and different integral inequalities. Another reason may be that the solution of the IVP can be written in the form a simple Volterra integral equation, whereas the solution of the BVP appears as a mixed (Volterra and Fredholm) integral equation.

8. Fractional-order impulsive differential equations

An impulsive differential equation is a special class of differential equation used to describe real-world phenomena more accurately, including evolutionary processes characterised by abrupt changes of the state at certain instants. In the literature two familiar impulses are found: instantaneous impulses and non-instantaneous impulses. In the case of instantaneous impulses, the time interval of the changes is relatively short in comparison to the total duration of the process, while in the non-instantaneous case, an impulsive action starts at an arbitrary point in time and remains active for a finite time interval. For details on the theory of the impulsive differential equation see the monographs by Lakshmikantham [38], Bainov [8] and Wang [74]. In 2013, Hernández [28] introduced a new class of impulsive differential equation with non-instantaneous impulses and studied the existence of a mild solution. Agarwal et al. [3] analysed a Caputo FDE with non-instantaneous impulses. For a detailed survey on non-instantaneous impulses on integer- and FDEs, we refer to the monograph by Agarwal et al. [4].

Wang et al. [81] studied the existence of the solution and the Ulam-Hyers stability of non-linear impulsive FDEs with Caputo derivative on the finite interval J = [0, T]:

$${}_{C}D_{0,t}^{\alpha}u(t) = f(t, u(t)), \quad t \in J' := J \setminus \{t_{1}, \dots, t_{m}\}, \quad 0 < \alpha < 1,$$

$$\Delta u(t_{k}) := u(t_{k}^{+}) - u(t_{k}^{-}) = I_{k}(u(t_{k}^{-})), \quad k = 1, 2, \dots, m,$$

$$u(0) = u_{0},$$

$$(8.1)$$

where $f: J \times \mathbb{R} \to \mathbb{R}$ is jointly continuous, $I_k : \mathbb{R} \to \mathbb{R}$ and t_k , k = 1, 2, ..., m, satisfy $0 = t_0 < t_1 < t_2 < \cdots < t_m < t_{m+1} = T$, $u(t_k^+) = \lim_{\epsilon \to 0^+} u(t_k + \epsilon)$ and $u(t_k^-) = \lim_{\epsilon \to 0^-} u(t_k + \epsilon)$ represent the right and left limits of u(t) at $t = t_k$. They established the results by using the fixed point theorem on a Banach space $PC(J, \mathbb{R}) = \{u : J \to \mathbb{R} : u \in C((t_k, t_{k+1}], \mathbb{R}), k = 0, 1, ..., m; u(t_k^+), u(t_k^-) \text{ exist} \}$

with $u(t_k) = u(t_k^-)$ } endowed with the norm

$$||u||_{PC} = \sup_{t \in J} |u(t)|.$$

Lemma 8.1. [81] Let $u \in PC(J, \mathbb{R})$ satisfy the following inequality

$$|u(t)| \le a(t) + b \int_0^t (t - \tau)^{\alpha - 1} |u(\tau)| d\tau + \sum_{0 < t_k < t} \theta_k |u(t_k^-)|,$$

where a(t) is non-negative continuous and non-decreasing function on J and b, θ_k are non-negative constants. Then,

$$|u(t)| \le a(t) (1 + \theta E_{\alpha}(b\Gamma(\alpha)t^{\alpha}))^k E_{\alpha}(b\Gamma(\alpha)t^{\alpha}), \quad \text{for } t \in (t_k, t_{k+1}],$$

where $\theta = \max\{\theta_k : k = 1, 2, \dots m\}.$

Definition 8.1. [81] A function $u \in PC(J, \mathbb{R})$ is a solution of the impulsive FDE (8.1) if u satisfies

$$u(t) = \begin{cases} u_{0} + \frac{1}{\Gamma(\alpha)} \int_{0}^{t} (t - \tau)^{\alpha - 1} f(\tau, u(\tau)) d\tau, & \text{for } t \in [0, t_{1}], \\ u_{0} + I_{1}(u(t_{1}^{-})) + \frac{1}{\Gamma(\alpha)} \int_{0}^{t} (t - \tau)^{\alpha - 1} f(\tau, u(\tau)) d\tau, & \text{for } t \in (t_{1}, t_{2}], \\ u_{0} + I_{1}(u(t_{1}^{-})) + I_{2}(u(t_{2}^{-})) + \frac{1}{\Gamma(\alpha)} \int_{0}^{t} (t - \tau)^{\alpha - 1} f(\tau, u(\tau)) d\tau, & \text{for } t \in (t_{2}, t_{3}], \\ \vdots \\ u_{0} + \sum_{k=1}^{m} I_{k}(u(t_{k}^{-})) + \frac{1}{\Gamma(\alpha)} \int_{0}^{t} (t - \tau)^{\alpha - 1} f(\tau, u(\tau)) d\tau, & \text{for } t \in (t_{m}, T], \end{cases}$$

$$(8.2)$$

They assume the following assumptions to establish the main results:

(A9) For arbitrary $(t, u) \in J \times \mathbb{R}$, there exist $C_f, M_f > 0$ and $q_1 \in [0, 1)$ such that $|f(t, u)| \leq C_f |u|^{q_1} + M_f$.

(A10) For arbitrary $u \in \mathbb{R}$, there exist $C_I, M_I > 0$ and $q_2 \in [0, 1)$ such that $|I_k(u)| \leq C_I |u|^{q_2} + M_I$ $k = 1, 2, \ldots, m$.

(A11) There exists a constant $K_I^{(k)} > 0$ such that $|I_k(u_1) - I_k(u_2)| \le K_I^{(k)} |u_1 - u_2|$, for all $u_1, u_2 \in \mathbb{R}$ and k = 1, 2, ..., m.

Theorem 8.1. Let the assumptions (A9), (A10), (A11) hold and $f \in C(J \times \mathbb{R}, \mathbb{R})$ be a Lipschitz-continuous function with respect to the second variable with a Lipschitz constant L_f . Let $\varphi \in C(J, \mathbb{R}_+)$ be a non-decreasing function and there exists a

constant $\lambda_{\varphi} > 0$ such that $\frac{1}{\Gamma(\alpha)} \int_0^t (t - \tau)^{\alpha - 1} \varphi(\tau) ds \leq \lambda_{\varphi} \varphi(t)$ for each $t \in J$. If a function $v \in PC(J, \mathbb{R}_+)$ satisfies the inequalities

$$\begin{aligned}
&\left|_{C} \mathcal{D}_{0,t}^{\alpha} v(t) - f(t, v(t))\right| \leq \varphi(t), \quad t \in J' \\
&\left|\Delta v(t_k) - I_k(v(t_k^-))\right| \leq \varphi(t_k), \quad k = 1, 2, \dots, m,
\end{aligned} \tag{8.3}$$

then there exists a unique solution $u \in PC(J, \mathbb{R})$ of the impulsive FDE (8.1) such that

$$|v(t) - u(t)| \le (m + \lambda_{\varphi}) M^* \varphi(t), \quad \forall t \in J, \tag{8.4}$$

where $M^* = E_{\alpha}(L_f T^{\alpha}) (1 + K_I E_{\alpha}(L_f T^{\alpha}))^m$ and $K_I = \max\{K_I^{(k)} : k = 1, 2, ..., m\}$. In other words, the impulsive FDE BVP (8.1) is Ulam-Hyers-Rassias stable.

Ding [21] studied the Ulam-Hyers stability of delay FDEs with instantaneous impulses by using the Banach fixed point theorem and the abstract Grönwall inequality.

Wang et al. [82] investigated the existence of the solution and Ulam-type stability of non-linear FDEs with non-instantaneous impulses with a Caputo derivative on the finite interval J = [0, T]:

$${}_{C}D_{0,t}^{\alpha}u(t) = f(t, u(t)), \ t \in (t_{k}, s_{k}], \ k = 0, 1, \dots, m, \ 0 < \alpha < 1,$$

$$u(t) = g_{k}(t, u(t)), \ t \in (s_{k-1}, t_{k}], \ k = 1, 2, \dots, m,$$

$$u(0) = u_{0},$$

$$(8.5)$$

where $0 = t_0 < s_0 < t_1 < s_1 < \ldots < t_m < s_m = T$, $f: J \times \mathbb{R} \to \mathbb{R}$ is continuous, and $g_k: [s_{k-1}, t_k] \times \mathbb{R} \to \mathbb{R}$ is continuous for each $k = 1, 2, \ldots, m$, the so-called non-instantaneous impulses.

Definition 8.2. [82] A function $u \in PC(J, \mathbb{R})$ is a mild solution of the FDE (8.5) if u satisfies

$$u(t) = \begin{cases} u_0 + \frac{1}{\Gamma(\alpha)} \int_0^t (t - \tau)^{\alpha - 1} f(\tau, u(\tau)) d\tau, & \text{for } t \in [0, s_0], \\ g_k(t, u(t)), & \text{for } t \in (s_{k-1}, t_k], k = 1, 2, \dots, m, \\ g_k(t_k, u(t_k)) - \frac{1}{\Gamma(\alpha)} \int_0^{t_k} (t_k - \tau)^{\alpha - 1} f(\tau, u(\tau)) d\tau \\ + \frac{1}{\Gamma(\alpha)} \int_0^t (t - \tau)^{\alpha - 1} f(\tau, u(\tau)) d\tau, & \text{for } t \in (t_k, s_k], k = 1, 2, \dots, m. \end{cases}$$

$$(8.6)$$

They introduced the concepts of generalised Ulam-Hyers stability for the non-instantaneous impulsive FDE (8.5).

Let $\varepsilon>0$ and $\varphi\in PC(J,\mathbb{R}_+)$ be non-decreasing. Consider the following inequalities:

$$\begin{cases} \left| {}_{C} \mathcal{D}_{0,t}^{\alpha} v(t) - f(t, v(t)) \right| \le \varphi(t), & t \in (t_{k}, s_{k}], \ k = 0, 1, \dots, m, \\ \left| v(t) - g_{k}(t, v(t)) \right| \le \varepsilon, & t \in (s_{k-1}, t_{k}], \ k = 1, 2, \dots, m. \end{cases}$$
(8.7)

Definition 8.3. [82] The non-instantaneous impulsive FDE (8.5) is said to be generalised Ulam-Hyers stable with respect to (φ, ε) if there exists a constant L > 0 such that, for each solution $v \in PC(J, \mathbb{R})$ of the inequality (8.7) there exists a solution $u \in PC(J, \mathbb{R})$ of the FDE (8.5) with

$$|v(t) - u(t)| \le L(\varphi(t) + \varepsilon), \quad t \in J.$$

They assume the following assumptions to establish the main results:

(A12) $g_k \in C([s_{k-1}, t_k] \times \mathbb{R}, \mathbb{R})$ and there are positive constants L_{g_k} , k = 1, 2, ..., m such that $|g_k(t, u_1) - g_k(t, u_2)| \le L_{g_k} |u_1 - u_2|$, for each $t \in (s_{k-1}, t_k]$, and for all $u_1, u_2 \in \mathbb{R}$.

(A13) The function $\varphi \in C(J, \mathbb{R}_+)$ is a non-decreasing function. There exist $c_{\varphi} > 0$ and 0 such that

$$\left(\int_0^t (\varphi(\varsigma))^{1/p} d\varsigma\right)^p \le c_{\varphi}\varphi(t), \quad \forall t \in J.$$

Theorem 8.2. [82] Assume (A12) and (A13) hold, and let $f \in C(J \times \mathbb{R}, \mathbb{R})$ be Lipschitz-continuous with respect to the second variable with a Lipschitz constant L_f . If a function $v \in PC(J, \mathbb{R})$ satisfies the inequality (8.7). Then, there exists a unique solution $u \in PC(J, \mathbb{R})$ of the FDE (8.5) as given in equation (8.6) such that

$$|v(t) - u(t)| \le \frac{\left(\frac{2c_{\varphi}}{\Gamma(\alpha)} \left(\frac{1-p}{\alpha-p}\right)^{1-p} T^{\alpha-p} + 1\right)}{1-M} [\varphi(t) + \varepsilon] \quad t \in J, \tag{8.8}$$

provided that $M = \max\{M_1, M_2\} < 1$, and where

$$M_1 = \max \left\{ \frac{L_f c_{\varphi}}{\Gamma(\alpha)} \left(\frac{1-p}{\alpha-p} \right)^{1-\alpha} (s_k^{\alpha-p} + t_k^{\alpha-p}) + L_{g_k} \mid k = 0, 1, 2, \dots, m \right\},$$

$$M_2 = \max \left\{ \frac{L_f}{\Gamma(\alpha+1)} (s_k^{\alpha} + t_k^{\alpha}) + L_{g_k} \mid k = 1, 2, \dots, m \right\}.$$

Similarly, Shankar and Bora [60,61] established the Ulam-Hyers and generalised Ulam-Hyers stability of the non-instantaneous impulsive integro-differential equation involving a Caputo derivative by using the Banach fixed point theorem and applied the obtained results to fractional RLC circuits as an application.

9. Conclusions

We presented a brief survey and introduced the methods to deal with the Ulam-Hyers stability of fractional-order differential equations (FDEs). The survey covers recent contributions in this area for various classes of FDEs such as linear FDEs, non-linear FDEs, delay FDEs, fractional-order boundary value problems (BVP), and impulsive FDEs. We also established a connection between the Lyapunov and Ulam-Hyers stability for dynamical systems and pointed out that the Ulam-Hyers stability is more general than the Lyapunov stability.

From this survey, one can observe that most of the results on Ulam-Hyers stability for FDEs have been established on a bounded interval, and none of the authors have tried to estimate the best Ulam-Hyers constant even for linear FDEs. As this field has large relevance in practical applications of FDEs, both more work on the development of the theoretical framework and establishing solutions for concrete problems are needed.

Acknowledgments

RM acknowledges funding from the German Research Foundation (DFG, grants ME 1535/12-1 and ME 1535/22-1). CL acknowledges funding from the National Natural Science Foundation of China (grant 12271339).

References

- [1] R.P. Agarwal, M. Benchohra, S. Hamani, A survey on existence results for boundary value problems of nonlinear fractional differential equations and inclusions. Acta. Appl. Math. 109, 973-1033 (2010).
- [2] R.P. Agarwal, S. Hristova, D. O'Regan, Boundary value problems for fractional differential equations of caputo type and Ulam type stability: Basic concepts and study. Axioms 12, 1-16 (2023).
- [3] R. Agarwal, S. Hristova, D. O'Regan, Non-instantaneous impulses Caputo fractional differential equation. Fract. Calc. Appl. Anal., **20**, 595-622 (2017).
- [4] R. Agarwal, S. Hristova, D. O'Regan, Non-instantaneous Impulses in Differential Equations, Springer Nature, Switzerland (2017).

- [5] A. Ali, K. Shah, Y. Li, Topological degree theory and Ulam's stability analysis of a boundary value problem of fractional differential Equations, *Frontiers in Functional Equations and Analytic Inequalities*. Springer Nature, Switzerland, 73-92 (2019).
- [6] F.M. Al-Askar, C. Cesarano, W.W. Mohammed, Multiplicative Brownian motion stabilizes the exact stochastic solutions of the Davey-Stewartson equations. Symmetry, 14(2176), 1-12 (2022).
- [7] R.L. Bagley, R.A. Calico, Fractional order state equations for the control of viscoelastically damped structures. J. Guid. Contr. Dyn., 14, 304-3011 (1991).
- [8] D.D. Bainov, P.S. Simeonov, *Theory of Impulsive Differential Equations*. Series on Advances in Mathematics for Applied Sciences, Vol. 28. World Scientific, Singapore (1995).
- [9] O. Bénichou, G. Oshanin, A unifying representation of path integrals for fractional Brownian motions. J. Phys. A: Math. Theor., **57**(22), 1-23 (2024).
- [10] C. Benzarouala, C. Tunc, Hyers-Ulam-Rassias stability of fractional delay differential equations with Caputo derivative. Math. Meth. Appl. Sci., 47, 13499-13509 (2024).
- [11] Z. Bouazza, M.S. Souid, V. Rakočević, On Ulam-Hyers-Rassias stability of the boundary value problem of Hadamard fractional differential equations of variable order. Afr. Mat., 33, 1-17 (2022).
- [12] N. Brillouët-Beluot, J. Brzdek, K. Cieplinski, On some recent developments in Ulam's type stability. Abstr. Appl. Anal. **2012**(1), 1-41 (2012).
- [13] L. Cadariu, V. Radu, Fixed point methods for the generalized stability of functional equations in a single variable. Fixed Point Theory App., **2008**, 1-15 (2008).
- [14] C. Chen, L. Liu, Q. Dong, Existence and Hyers-Ulam stability for boundary value problems of multi-term Caputo fractional differential equations. Filomat, **37**, 9679-9692 (2023).
- [15] Y. Cheng, Z.B. Bai, Existence results of non-local integro-differential problem with singularity under a new fractional Musielak-Sobolev space. J. Phys. A: Math. Theor., 58(4), 1-28 (2025).
- [16] K. Ciepliński, Applications of fixed point theorems to the Hyers-Ulam stability of functional equations- a survey. Ann. Funct. Anal., 3, 151-164 (2012).
- [17] N.D. Cong, H.T. Tuan, Existence, uniqueness, and exponential boundedness of global solutions to delay fractional differential equations. 14(93), 1-12 (2017).
- [18] D.X. Cuong, On the Hyers-Ulam stability of Riemann-Liouville multi-order fractional differential equations. Afr. Mat., **30**, 1041-1047 (2019).
- [19] A. Das, R. Jain, H.K. Nashine, A fixed point result via new condensing operator and its application to a system of generalized proportional fractional integral equations. J. Pseudo-Differ. Oper. Appl., 14(21), 1-15 (2023).
- [20] F. Develî, O. Duman, Existence and stability analysis of solution for fractional delay differential equations. Filomat, 37, 1869-1878 (2023).
- [21] Y. Ding, Ulam-Hyers stability of fractional impulsive differential equations. J. Nonlinear Sci. Appl., 11, 953-959 (2018).
- [22] E.El-Hady, S. Öğrekci, On Hyers-Ulam-Rassias stability of fractional differential equations with Caputo derivative. J. Math. Computer Sci., 22, 325-332 (2021).
- [23] T. Erneux, Applied Delay Differential Equations. Springer, New York (2009).
- [24] C. Friedrich, Relaxation and retardation functions of the Maxwell model with fractional derivatives. Rheol. Acta, **30**, 151-158 (1991).
- [25] T. Guggenberger, A. Chechkin, R. Metzler, Fractional Brownian motion in superharmonic potentials and non-Boltzmann stationary distributions. J. Phys. A: Math. Theor.,

- **(54)**29LT01, 1-17 (2021).
- [26] O. Hatori, K. Kobayasi, T. Miura, H. Takagi, S.E. Takahasi, On the best constant of Hyers-Ulam stability. J. Nonlinear Convex Anal., 5, 387-393 (2004).
- [27] F. Haq, K. Shah, G. ur Rahman, M. Shahzad, Hyers-Ulam stability to a class of fractional differential equations with boundary conditions. Int. J. Appl. Comput. Math., 3, 1135-1147 (2017).
- [28] E. Hernández, D. O'Regan, On a new class of abstract impulsive differential equations. Proc. Am. Math. Soc., 141, 1641-1649 (2013).
- [29] R. Hilfer, Applications of Fractional Calculus in Physics. World Scientific, Singapore (2000).
- [30] S. Hristova, M.I. Abbas, Ulam type stability analysis for generalized proportional fractional differential equations. Carpathian Math. Publ., **16**(1), 114-127 (2024).
- [31] D.H. Hyers, On the stability of the linear functional equation. Proce. Natl. Acad. Sci. USA, 27, 222-224 (1941).
- [32] D.H. Hyers, G. Isac, Th.M. Rassias, Stability of Functional Equations in Several Variables. Birkhäuser, Basel (1998).
- [33] S.M. Jung, D. Popa, Th.M. Rassias, On the stability of the linear functional equation in a single variable on complete metric gropus. J. Glob. Optim., **59**, 165-171 (2014).
- [34] H.K. Khalil, Nonlinear Systems. Prentice Hall, Englewood Cliffs (1996).
- [35] A. N. Kolmogorov, Wienersche Spiralen und einige andere interessante Kurven im Hilbertschen Raum, C. R. (Doklady) Acad. Sci. URSS (N.S.) **26**, 115-118 (1940).
- [36] C.P. Li, M. Cai, Theory and Numerical Approximations of Fractional Integrals and Derivatives. SIAM, Philadelphia (2019).
- [37] Y. Li, Y.Q. Chen, I. Podlubny, Stability of fractional-order nonlinear dynamic systems: Lyapunov direct method and generalized Mittag-Leffler stability. Comput. Math. Appl., 59(5), 1810-1821 (2010).
- [38] V. Lakshmikantham, D.D. Bainov, P.S. Simeonov, *Theory of Impulsive Differential Equations*. Series in Modern Applied Mathematics, Vol. 6. World Scientific, Singapore (1989).
- [39] M. Lakshmanan, D.V. Senthilkumar, Dynamics of Nonlinear Time-Delay Systems. Springer, New York (2011).
- [40] C.P. Li, F.R. Zhang, A survey on the stability of fractional differential equations. Eur. Phys. J. Spec. Top., 193, 27-47 (2011).
- [41] H. Liu, Y. Li, Hyers-Ulam stability of linear fractional differential equations with variable coefficients. Adv. Differ. Equ., **2020**(404), 1-10 (2020).
- [42] F. Mainardi, Fractional calculus and waves in linear viscoelasticity. World Scientific, Singapore (2010).
- [43] B.B. Mandelbrot, J.W. van Ness, Fractional Brownian motions, fractional noises and applications. SIAM Rev., **10**(4), 422-437 (1968).
- [44] A. Manapany, S. Fumeron, M. Henkel, Fractional diffusion equations interpolate between damping and waves. J. Phys. A: Math. Theor., **57**(35), 355202 (2024).
- [45] F. Masood, C. Cesarano, O. Moaaz, S.S. Askar, A.M. Alshamrani, H. El-Metwally, Kneser-Type oscillation criteria for half-Linear delay differential equations of third order. Symmetry, 15(1994), 1-18 (2023).
- [46] B. Meerson, P.V. Sasorov, Fractional Brownian motion in confining potentials: non-equilibrium distribution tails and optimal fluctuations. J. Phys. A: Math. Theor., **57**(44), 445002 2024.
- [47] R. Metzler, J. Klafter, The random walk's guide to anomalous diffusion: A fractional dynamics

- approach. Phys. Rep., **339**(1), 1-77 (2000).
- [48] R. Metzler, J. Klafter, The restaurant at the end of the random walk: recent developments in fractional dynamics descriptions of anomalous dynamical processes. J. Phys. A: Math. Theor., 37, R161-R208 (2004).
- [49] R. Metzler, J. Klafter, Accelerating Brownian motion: A fractional dynamics approach to fast diffusion, Europhys. Lett., 51(5), 492-498 (2000).
- [50] O. Moaaz, C. Cesarano, A. Muhib, Some new oscillation results for fourth-order neutral differential equations. Eur. J. Pure Appl. Math., 13(2), 185-199 (2020).
- [51] T.F. Nonnenmacher, W.G. Glöckle, A fractional model for mechanical stress relaxation. Philos. Mag. Lett., **64**, 89-93 (1991).
- [52] M. Onitsuka, T. Shoji, Hyers-Ulam stability of first-order homogeneous linear differential equations with a real-valued coefficient. Appl. Math. Lett., **63**, 102-108 (2017).
- [53] H.K. Pathak, An Introduction to Nonlinear Analysis and Fixed Point Theory. Springer Nature, Singapore (2017).
- [54] I. Podlubny, Fractional Differential Equations. Academic Press, San Diego (1999).
- [55] D. Popa, I. Rasa, On the stability of some classical operators from approximation theory. Expo. Math., 31, 205-214 (2013).
- [56] Th.M. Rassias, On the stability of the linear mapping in Banach spaces. Proc. Arner. Math. Soc., 72, 297-300 (1978).
- [57] D.A. Refaai, M.M.A. El-Sheikh, G.A.F. Ismail, M. Zakarya, G. AlNemer, H.M. Rezk, Stability of nonlinear fractional delay differential equations. Symmetry 14, 1-9 (2022).
- [58] I.A. Rus, Remarks on Ulam stability of the operatorial equations. Fixed Point Theory, **10**, 305-320 (2009).
- [59] H. Salah, M. Anis, C. Cesarano, S.S. Askar, A.M. Alshamrani, E.M. Elabbasy, Fourth-order differential equations with neutral delay: Investigation of monotonic and oscillatory features. AIMS Mathematics, 9(12), 34224-34247 (2024).
- [60] M. Shankar, S.N. Bora, Generalized Ulam-Hyers-Rassias stability of solution for the Caputo fractional non-instantaneous impulsive integro-differential equation and its application to fractional RLC circuit. Circuits, Systems, and Signal Processing, 42, 1959-1983 (2023).
- [61] M. Shankar, S.N. Bora, Ulam-Hyers stability of non-instantaneous impulsive integrodifferential equation of real-order with Caputo derivative with application to circuits. J. Nonlinear Evol. Equ. Appl., 2024, 45-65 (2024).
- [62] Y. Shen, W. Chen, Laplace transform method for the Ulam stability of linear fractional differential equations with constant coefficients. Mediterr. J. Math., 14(1), 25 (2017).
- [63] L.J. Slater, Generalized Hypergeometric Functions. Cambridge University Press, Cambridge (1966).
- [64] S. Subashmoorthy, P. Balasubramaniam, Hyers-Ulam-Rassias stability results for some nonlinear fractional integral equations using the Bielecki metric. Math. Meth. Appl. Sci., 47, 11201-11214 (2024).
- [65] H.G. Sun, Y. Zhang, D. Baleanu, Y.Q. Chen, A new collection of real world applications of fractional calculus in science and engineering. Commun. Nonlinear Sci. Numer. Simulat., 64, 213-231 (2018).
- [66] H. Takagi, T. Miura, S.E. Takahasi, Essential norms and stability constants of weighted composition operators on C(X). Bull. Korean Math. Soc., **40**, 583-591 (2003).
- [67] M.S. Tavazoie, M. Haeri, Stabilization of unstable fixed points of chaotic fractional order systems by a stable fractional PI controller. Eur. J. Control, 14, 247-257 (2008).

- [68] J.A. Tenereiro Machado, Analysis and design of fractional-order digital control systems. Syst. Anal. Model. Simul., 27, 107-122 (1997).
- [69] N.T. Thanh, H. Trinh, V.N. Phat, Stability analysis of fractional differential time-delay equations. IET Control Theory Appl., 11(7), 1006-1015 (2017).
- [70] S.M. Ulam, Problems in Modern Mathematics. Wiley, New York (1960).
- [71] A. Valov, B. Meerson, Dynamical large deviations of the fractional Ornstein-Uhlenbeck process, J. Phys. A: Math. Theor., **58**, 095002 (2025).
- [72] M. Vidyasagar, *Nonlinear System Analysis*. Prentice Hall, Englewood Cliffs, New Jersey (1978).
- [73] H. Vu, J.M. Rassias, N.V. Hoa, Hyers-Ulam stability for boundary value problem of fractional differential equations with κ -Caputo fractional derivative. Math. Meth. Appl. Sci., **46**, 438-460 (2023).
- [74] J. Wang, M. Fečkan, Y. Zhou, A survey on impulsive fractional differential equations. Fract. Calc. Appl. Anal., 19, 806-831 (2016).
- [75] J. Wang, M. Fĕckan, Y. Zhou, Presentation of solutions of impulsive fractional Langevin equations and existence results. Eur. Phys. J. Special Top., 222, 1855-1872 (2013).
- [76] J.R. Wang, X.Z. Li, A uniform method to Ulam-Hyers stability for some linear fractional equations. Mediterr. J. Math., 13, 625-635 (2016).
- [77] J.R. Wang, L.L. Lv, Y. Zhou, New concepts and results in stability of fractional differential equations. Commun. Nonlinear Sci. Numer. Simul., 17, 2530-2538 (2012).
- [78] J. Wang, L. Lv, Y. Zhou, Ulam stability and data dependence for fractional differential equations with Caputo derivative. Electron J. Qualit. Th. Diff. Equat., 63, 1-10 (2011).
- [79] C. Wang, T.Z. Xu, Hyers-Ulam stability of a class of fractional linear differential equations. Kodai Math. J., 38, 510-520 (2015).
- [80] C. Wang, T.Z. Xu, Hyers-Ulam stability of fractional linear differential equations involving Caputo fractional derivatives. Applications of Math., 60, 383-393 (2015).
- [81] J. Wang, Y. Zhou, M. Fečkan, Nonlinear impulsive problems for fractional differential equations and Ulam stability. Comput. Math. Appl., **64**, 3389-3405 (2012).
- [82] J. Wang, Y. Zhou, Z. Lin, On a new class of impulsive fractional differential equations. Appl. Math. Comput., 242, 649-657 (2014).
- [83] H. Ye, J. Gao, Y. Ding, A generalized Gronwall inequality and its application to a fractional differential equation. J. Math. Anal. Appl., 328, 1075-1081 (2007).